# SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

## FORM 8-K

# CURRENT REPORT PURSUANT TO SECTIONS 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (date of earliest event reported): November 1, 2010

## UNIVERSAL ELECTRONICS INC.

(Exact name of Registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 0-21044

(Commission File No.)

33-0204817 (I.R.S. Employer Identification No.)

6101 Gateway Drive Cypress, California 90630 (Address of principal executive offices, with Zip Code)

(714) 820-1000 (Registrant's telephone number, including area code):

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

## **TABLE OF CONTENTS**

<u>Item 1.01 Entry into a Material Definitive Agreement</u>

Item 2.01 Completion of Acquisition or Disposition of Assets

Item 2.02 Results of Operations and Financial Condition

Item 3.02 Unregistered Sales of Equity Securities

Item 7.01 Regulation FD Disclosure

Item 9.01 Financial Statements and Exhibits

**SIGNATURES** 

EX-2.1

EX-99.1

EX-99.2

## Item 1.01 Entry into a Material Definitive Agreement.

Universal Electronics Inc. (the "Company" or "UEI") and U.S. Bank National Association, have entered into an Amended and Restated Credit Agreement, dated as of November 1, 2010. The amendments added a new \$35.0 million secured term loan facility ("Term Loan") for the purpose of financing a portion of the Company's acquisition of Enson Assets Limited. In addition, the Company's existing \$15.0 million unsecured revolving credit line with U.S. Bank ("Credit Facility") became a secured facility, was increased to \$20.0 million, and the expiration date of the Credit Facility was extended from October 31, 2011 to November 1, 2012.

The Company's new Term Loan may only be utilized to finance the acquisition of Enson Assets Limited and to pay related transaction costs, fees, and expenses. The Term Loan is subject to certain quarterly financial covenants related to the Company's cash flow, fixed charges, quick ratio, and net income. The minimum principal payments for the Term Loan are \$2.2 million each. Principal and interest payments are payable on January 5, April 5, July 5, and October 5 of 2011. In addition, a final payment equal to the unpaid principal balance plus accrued interest is due on the Term Loan maturity date. The Term Loan maturity date is November 1, 2011. Amounts paid or prepaid on the Term Loan may not be re-borrowed.

As security for the payment and performance of all obligations under the Term Loan and the Credit Facility, the Company has pledged to U.S. Bank 65% of its ownership in Enson Assets Limited as collateral.

Under the Term Loan the Company may elect to pay interest based on the bank's prime rate or LIBOR plus a fixed margin of 1.5%. The applicable LIBOR (1, 3, 6, or 12-month LIBOR) corresponds with the loan period we select. On November 1, 2010, the 1-month LIBOR plus the fixed margin was approximately 1.8% and the bank's prime rate was 3.25%. If a LIBOR rate loan is prepaid prior to the completion of the loan period, the Company must pay the bank the difference between the interest the bank would have earned had prepayment not occurred and the interest the bank actually earned.

In addition, Item 2.01 of this Current Report on Form 8-K is hereby incorporated into this Item 1.01 by reference.

## Item 2.01 Completion of Acquisition or Disposition of Assets.

## Summary

On November 3, 2010, the Company's subsidiary, UEI Hong Kong Private Limited, entered into an agreement to acquire all of the issued shares in the capital of Enson Assets Limited, a limited liability company organized under the Laws of the British Virgin Islands, for total consideration of approximately \$125.4 million, in cash and the Company's common stock. This acquisition was consummated pursuant to a Stock Purchase Agreement, dated as of November 3, 2010, among Universal Electronics Inc., UEI Hong Kong Private Limited and CG International Holdings Limited ("CGI"), a closely-held exempted company incorporated in the Cayman Islands. The Company has not yet estimated the fair value of the contingent payments, which may change the total purchase price of \$125.4 million. A total of \$5.0 million of the purchase price was held back at the closing and will be applied to any CGI payments required as a result of the failure to meet both a net asset target and an earnings target for Enson Assets Limited and its subsidiaries ("Enson") (see "Contingent Consideration" below).

The Purchase Agreement includes certain covenants, representations and warranties which are customary in transactions of this type, including certain indemnification obligations of the parties with respect to breaches of covenants, representations and warranties, subject to certain springing baskets, maximum liability and time limitations. CGI and its majority shareholder have undertaken non-competition and

non-solicitation obligations relating to the customers and employees of Enson that will be effective for a period of two years from the Closing Date. The representations, warranties and covenants contained in the Purchase Agreement including, without limitation, the Disclosure Schedule referenced in the Purchase Agreement, were made only for purposes of the Purchase Agreement and were solely for the benefit of the parties to the Purchase Agreement, and may be subject to limitations agreed upon by the contracting parties, including being qualified by confidential disclosures exchanged between the parties in connection with the execution of the Purchase Agreement. The representations and warranties may have been made for the purpose of allocating contractual risk between the parties to the Purchase Agreement instead of establishing these matters as facts, and may be subject to standards of materiality or knowledge applicable to the contracting parties that differ from those applicable to investors. Investors should not rely upon the representations, warranties, covenants or Disclosure Schedule, or any descriptions thereof as characterizations of the actual state of facts or condition of Enson, the Company or any of their respective subsidiaries or affiliates. Moreover, information concerning the subject matter of the representations and warranties may change after the date of the Purchase Agreement, which subsequent information may or may not be fully reflected in the registrant's public disclosures.

## **Enson Description**

Enson is a leading manufacturer of remote controls. It is one of the Company's significant suppliers. The Enson corporate office, located in Hong Kong, is approximately 6,000 square feet and employs 53 people. Enson controls two factories located in the Peoples Republic of China.

The southern factory is located in Guang Dong Province, Peoples Republic of China within the city of Guang Zhou. The Guang Zhou factory is approximately 710,203 square feet and employs 796 people, with an additional 4,916 factory workers contracted through an agency agreement.

The northern factory is located in Jiang Su Province, Peoples Republic of China within the city of Yang Zhou. The Yang Zhou factory is approximately 768,543 square feet and employs 384 people, with an additional 4,415 factory workers contracted through an agency agreement.

During the three months ended September 30, 2010 and 2009, the amount of components and finished goods the Company purchased from Enson composed the following:

Three Months Ended September 30,				
2010		2009	2009	
	% of Total		% of Total	
	Inventory		Inventory	
\$ (thousands)	Purchases	\$ (thousands)	Purchases	
10,173	22.0%	11,436	22.8%	

During the nine months ended September 30, 2010 and 2009, the amount of components and finished goods the Company purchased from Enson composed the following:

Nine Months Ended September 30,				
2010	2010		2009	
-	% of Total		% of Total	
	Inventory		Inventory	
\$ (thousands)	Purchases	\$ (thousands)	Purchases	
28,763	20.5%	32,926	23.1%	

The total accounts payable to Enson on September 30, 2010 and December 31, 2009 were the following:

	September	September 30, 2010		December 31, 2009	
		% of Accounts		% of Accounts	
	\$ (thousands)	Payable	\$ (thousands)	Payable	
Enson	9,103	29.4%	11,887	30.1%	

Page 3

#### **Primary Reasons for the Acquisition**

The value the Company anticipates from this acquisition relates primarily to the following:

- Enson should increase the Company's market position in the strategically important consumer electronics market with its historic strength with leading Japanese customers. The Company has not historically been well positioned in this market.
- Enson currently produces approximately one-third of the Company's volume, therefore, the Company may decrease third party supplier purchases. In addition, Enson has available manufacturing capacity, which may provide it the ability to increase utilization of its existing factories.
- The Company may utilize Enson's in-place management and personnel to assist in implementing its plan to place more operations, logistics, quality, program management, engineering, sales, and marketing personnel in the Asia region.
- Enson's full line of remotes, from dedicated to higher-end universal, should assist the Company to further penetrate the growing Asian and Latin
  American subscription broadcasting markets. The lower subscriber revenue in these markets can cause them to begin with lower-cost dedicated
  remotes and to later transition to universal remote controls.
- Acquiring Enson should allow the Company to gain purchasing economies.

#### Consideration

Under the terms of the agreement, the Company paid aggregate consideration of approximately \$125.4 million (see "Contingent Consideration" below). The consideration transferred consisted of \$95.0 million in cash and 1,460,000 of newly issued shares of the Company's common stock. These shares were issued in a transaction that was intended to be exempt from registration under the Securities Act of 1933, as amended, under the provisions of Regulation S of the Securities Act of 1933, as amended. The sources of the consideration were approximately the following:

(In thousands) Source Description	Amount	Percentage of Consideration
Existing cash and cash equivalents	\$ 54,000	43.1%
Funds from new Term Loan	35,000	27.9%
Funds from Credit Facility	6,000	4.8%
Newly issued shares of Universal Electronics, Inc. common stock	30,412	24.2%
	\$125,412	100.0%

## Liquidation of the Company's Term Deposit

On July 2, 2010, the Company entered into a six-month term deposit cash account at Wells Fargo Bank denominated in Hong Kong dollars. The deposit amount related to this account was \$49.5 million, and was earning interest at an annual rate of 0.5%. The term was set to end on December 30, 2010. On October 28, 2010, the Company elected to liquidate this term deposit account to assist the Company with the funding of the acquisition of Enson.

## New Secured Credit Agreement

On November 1, 2010, the Company amended and restated its existing credit agreement with U.S. Bank. The amendments added a new \$35.0 million secured Term Loan for the purpose of financing a portion of the Company's acquisition of Enson. In addition, the Company's existing \$15.0 million unsecured Credit Facility became a secured facility, was increased to \$20.0 million, and the expiration date was extended from October 31, 2011 to November 1, 2012.

The Company's new Term Loan may only be utilized to finance the acquisition of Enson and to pay related transaction costs, fees, and expenses. The Term Loan is subject to certain quarterly financial covenants related to the Company's cash flow, fixed charges, quick ratio, and net income. The minimum principal payments for the Term Loan are \$2.2 million each. Principal and interest payments are payable on January 5,

April 5, July 5, and October 5 of 2011. In addition, a final payment equal to the unpaid principal balance plus accrued interest is due on the Term Loan maturity date. The Term Loan maturity date is November 1, 2011. Amounts paid or prepaid on the Term Loan may not be re-borrowed.

As security for the payment and performance of all obligations under the Term Loan and Credit Facility, the Company has pledged to U.S. Bank 65% of the Company's ownership in Enson Assets Limited as collateral.

Under the Term Loan, the Company may elect to pay interest based on the bank's prime rate or LIBOR plus a fixed margin of 1.5%. The applicable LIBOR (1, 3, 6, or 12-month LIBOR) corresponds with the loan period we select. On November 1, 2010, the 1-month LIBOR plus the fixed margin was approximately 1.8% and the bank's prime rate was 3.25%. If a LIBOR rate loan is prepaid prior to the completion of the loan period, the Company must pay the bank the difference between the interest the bank would have earned had prepayment not occurred and the interest the bank actually earned.

## **Contingent Consideration**

Net Asset Target on November 3, 2010

Enson will deliver to the auditors for review a statement of net assets of Enson as of the closing date and the auditors will confirm the amounts in a report. To the extent that the Enson net assets are less than \$68.5 million, CGI will pay to the Company an amount equal to the difference, plus interest. Interest will be calculated as of the closing date using the prime rate as reported in The Wall Street Journal on the day of this determination. The Company believes that the holdback funds will be adequate to cover any repayment, if required.

To the extent that the Enson net assets are greater than \$68.5 million, the Company will pay CGI an amount equal to the difference, plus interest. Interest will be calculated as of the closing date using the prime rate as reported in The Wall Street Journal on the day of this determination. The Company does not believe any payment, if required, will be material.

Earnings Target for the Twelve Months Ending March 31, 2011

To the extent that Enson's earnings for the year ended March 31, 2011 are less than \$16.2 million, CGI will pay the Company back an amount equal to the product of (a) the difference between Enson's earnings and \$16.2 million, multiplied by (b) one and one half, plus interest. Interest will be calculated as of the closing date using the prime rate as reported in The Wall Street Journal on the date of this determination. CGI is required to make this payment within five business days of the issuance of the auditor's report on Enson's accounts.

For the purposes of this calculation, Enson's earnings is defined as Enson's consolidated profit before tax for the twelve months ending March 31, 2011 excluding certain agreed upon adjustments, including without limitation, the following items: profit related to UEIC sales, investment income, other income, other expenses, other gains and losses and interest expenses.

Seven months of the year ended March 31, 2011 are complete and Enson appears to be on track to meet this \$16.2 million earnings target.

## **Purchase Accounting**

Currently, the Company is performing the preliminary purchase accounting analysis, which requires the Company to recognize the assets acquired and liabilities assumed at the acquisition date, measured at their fair values. The Company anticipates that it will arrive at the preliminary purchase accounting during the fourth quarter of 2010. The preliminary purchase accounting will be based upon preliminary estimates and assumptions that are subject to change within the purchase measurement period (generally one year from the acquisition date to reflect new information about facts and circumstances existing on the acquisition date).

Although the Company believes the Enson transaction will be accretive in the first year and grow more significantly in the long term, most acquisitions of this type involve the purchase of significant limited life intangible assets which typically result in substantial amortization charges. In addition, goodwill may be recognized associated with this transaction which is subject to regular impairment testing. There can be no assurance that the integration will be successful or that the customer bases, products or technologies will generate sufficient revenue to offset the associated costs or effects.

## **Acquisition Costs**

The Company expects the total acquisition related costs for the Enson transaction to range between \$0.8 million and \$1.2 million. Of this amount \$0.1 million was expensed during the three months ended September 30, 2010. The Company expects to incur the remaining acquisition costs primarily during the three months ended December 31, 2010. These costs are included, and will be included, in selling, general and administrative expenses in accordance with U.S. GAAP.

The foregoing description of certain terms of the Purchase Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Purchase Agreement, a copy of which is attached as Exhibit 2.1 to this Current Report on Form 8-K.

#### Item 2.02 Results of Operations and Financial Condition

On November 4, 2010, Universal Electronics Inc. is issuing a press release and holding a conference call regarding its financial results for the third quarter 2010 and other matters. A copy of the press release is included as Exhibit 99.1 to this report.

Pursuant to General Instruction B2 of Form 8-K, the information contained in Exhibit 99.1 will be deemed furnished, and not "filed," for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or incorporated by reference in any filing under the Exchange Act or the Securities Act of 1933, as amended, except as expressly set forth by specific reference in any such filing.

## Item 3.02 Unregistered Sales of Equity Securities.

Item 2.01 of this Current Report on Form 8-K is hereby incorporated into this Item 3.02 by reference.

Pursuant to the Purchase Agreement, 1,460,000 newly issued shares of Universal Electronics Inc. ("UEI") common stock, par value \$.01 per share, were delivered as a portion of the purchase price under the Purchase Agreement. The shares were issued in reliance upon an exemption from registration under the Securities Act, pursuant to Regulation S promulgated under the Securities Act. UEI and UEI Hong Kong Private Limited complied with the conditions of Rule 903 of Regulation S, including, but not limited to, the following: (i) CGI is not a U.S. person and was offered and sold its shares in accordance with the provisions of Regulation S; (ii) an appropriate legend is required to be affixed to the shares in accordance with Regulation S; (iii) CGI has represented that it is not acquiring the shares for the account or benefit of a U.S. person; (iv) CGI has agreed to resell the shares only in accordance with the

provisions of Regulation S, pursuant to a registration statement under the Securities Act of 1933, as amended or pursuant to an available exemption from registration; and (v) CGI has agreed not to engage in hedging activities involving UEI's common stock. In the Purchase Agreement, CGI has acknowledged that UEI will implement the restrictions on transfer contained in the Purchase Agreement, which preclude any transfer of the shares which is not made in accordance Regulation S, not registered under the Securities Act, or not made pursuant another available exemption.

## **Item 7.01 Regulation FD Disclosure**

On November 4, 2010, Universal Electronics Inc. is issuing a press release and holding a conference call regarding its acquisition of Enson Assets Limited. A copy of the press release is included as Exhibit 99.2 to this report.

Pursuant to General Instruction B2 of Form 8-K, the information contained in Exhibit 99.2 will be deemed furnished, and not "filed," for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or incorporated by reference in any filing under the Exchange Act or the Securities Act of 1933, as amended, except as expressly set forth by specific reference in any such filing.

## **Item 9.01 Financial Statements and Exhibits**

(a) Financial Statements of Businesses Acquired

The financial statements of Enson Assets Limited required to be filed pursuant to Item 9.01(a) have been omitted as permitted by Item 9.01(a)(4). The Company anticipates that such financial statements will be filed on or before January 17, 2011.

(b) Pro Forma Financial Information

The pro forma financial information required to be filed pursuant to Item 9.01(b) has been omitted as permitted by Item 9.01(b)(2). UEI anticipates that such financial statements will be filed on or before January 17, 2011.

- (c) Exhibits. The following exhibits are furnished with this report.
- 2.1 Stock Purchase Agreement dated as of November 3, 2010, among Universal Electronics Inc., UEI Hong Kong Private Limited and CG International Holdings Limited\*
- 99.1 Press Release of Universal Electronics Inc. dated November 4, 2010\*\*
- 99.2 Press Release of Universal Electronics Inc. dated November 4, 2010\*\*

<sup>\*</sup> Attachments to the Purchase Agreement, identified on Exhibit 2.1, have been omitted as permitted by Item 601(b)(2) of Regulation S-K. UEI hereby undertakes to furnish supplementally to the Securities and Exchange Commission a copy of any omitted attachment upon request.

<sup>\*\*</sup> Pursuant to General Instruction B2 of Form 8-K, the information contained in Exhibits 99.1 and 99.2 will be deemed furnished, and not "filed," for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or incorporated by reference in any filing under the Exchange Act or the Securities Act of 1933, as amended, except as expressly set forth by specific reference in any such filing.

## **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Universal Electronics Inc.

Date: November 4, 2010 By: /s/ Bryan Hackworth

Bryan Hackworth Chief Accounting Officer (Principal Financial Officer)

Page 8

## STOCK PURCHASE AGREEMENT

By and Among

## CG INTERNATIONAL HOLDINGS LIMITED,

## UEI HONG KONG PRIVATE LIMITED

And

UNIVERSAL ELECTRONICS INC.

For the purchase of all issued shares in the capital of Enson Assets Limited, a limited liability company organized under the Laws of the British Virgin Islands

Dated as of November 3, 2010

1

ARTICLE I PURCHASE AND SALE OF SHARES	1
1.1 Purchase and Sale of Shares; Closing	1
1.2 Purchase Price	1
1.3 Net Asset Adjustment	3
1.4 Earnings Adjustment	4
ARTICLE II REPRESENTATIONS AND WARRANTIES	5
2.1 Representations and Warranties of Seller	5
2.2 Stock Portion Related Representations and Warranties	15
2.3 Representations and Warranties of Buyer	17
2.4 Representations and Warranties of UEI	18
2.5 Further Assurances by Buyer and UEI	20
ARTICLE III COVENANTS	20
3.1 Closing	20
3.2 Confidentiality	20
3.3 Further Assurances	20
3.4 Maintenance of Existence; Adequate Funds	21
3.5 Consents	21
3.6 Transfer and Reorganization Taxes	21
ARTICLE IV CERTAIN ADDITIONAL COVENANTS	21
4.1 Expenses	21
4.2 Press Releases and Disclosure	21
4.3 Government and other Regulatory Approvals	21
4.4 Reporting Requirements and Furnishing of Information	22
4.5 Piggy-back Registrations	22
4.6 Post-Closing Operation of the Company	23
4.7 Transfer of Shares to shareholders of Seller	23
4.8 Removal of Restricted Legend	23
ARTICLE V CLOSING DELIVERIES	23
5.1 Seller's Deliveries	23
5.2 Buyer's Deliveries	25
ARTICLE VI [INTENTIONALLY OMITTED]	26
ARTICLE VII SURVIVAL; INDEMNIFICATION	26
7.1 Survival of Provisions; Assertion of Claims	26

7.2 Indemnity Obligations	26
7.3 Limitations on Indemnification	27
7.4 Notice of Claims	27
7.5 Third Party Claims	27
7.6 Objection to Claim Notice	28
7.7 Payment	29
7.8 No Limitation on Rights to Indemnification	29
ARTICLE VIII GENERAL	30
8.1 Governing Law	30
8.2 Schedules, Addenda and Exhibits	30
8.3 Amendments	30
8.4 Entire Agreement	30
8.5 Assignment	30
8.6 Counterparts	30
8.7 Waivers	30
8.8 Third Parties	30
8.9 Headings	30
8.10 Gender and Number; Section and Article References	30
8.11 Time of Essence	31
8.12 Interpretation	31
8.13 Notices	31
8.14 Process Agents	32
ii	

## TABLE OF EXHIBITS

Exhibit A

Exhibit B

[Intentionally omitted.]
Form of Non-Competition Agreement
Form of Key Employee Employment Agreement
Form of Certificate by Officer Exhibit C

Exhibit D

## STOCK PURCHASE AGREEMENT

THIS STOCK PURCHASE AGREEMENT (this "<u>Agreement</u>"), dated November 3, 2010, is entered into by and among CG International Holdings Limited, an exempted company incorporated in the Cayman Islands under the Companies Law with limited liability ("<u>Seller</u>"), UEI Hong Kong Private Limited, a company organized under the Laws of Hong Kong ("<u>Buyer</u>") and Universal Electronics Inc., a Delaware corporation (for the limited purposes expressly set forth herein) ("<u>UEI</u>"). Seller, Buyer and UEI are each individually referred to herein as a "<u>Party</u>" and collectively as the "<u>Parties</u>."

WHEREAS, Buyer desires to purchase from Seller, and Seller desires to sell to Buyer, the Shares (as defined herein), upon the terms and subject to the conditions set forth in this Agreement, which collectively will constitute all of the issued share capital of Enson Assets Limited, a limited liability company organized under the Laws of the British Virgin Islands ("Enson") as at the Closing; and

WHEREAS, certain capitalized terms used herein and not otherwise defined shall have the meanings ascribed to such terms in <u>Annex A</u> attached hereto.

NOW, THEREFORE, in consideration of the Parties' mutual covenants and agreements set forth herein, and for other good, valuable and adequate consideration received, the Parties agree as follows:

#### ARTICLE I

## PURCHASE AND SALE OF SHARES

## 1.1 Purchase and Sale of Shares; Closing.

- (a) <u>Purchase and Sale of Shares</u>. Subject to the terms and conditions of this Agreement, at the Closing, Seller shall transfer and deliver to Buyer, and Buyer shall purchase and accept from Seller, all of the right, title and interest of Seller, as of the Closing Date, in and to the Shares.
- (b) <u>Closing</u>. The consummation of the transactions contemplated by this Agreement (the "<u>Closing</u>") shall take place at 10:00 A.M. US time as of the date hereof (the "<u>Closing Date</u>"). Legal title, equitable title and risk of loss with respect to the Shares shall not pass to Buyer until the Shares are transferred at the Closing, which transfer, once it has occurred, shall be deemed effective for tax, accounting and other computational purposes as of 11:59 P.M. US time on the Closing Date.
- **1.2** <u>Purchase Price</u>. Subject to the terms and conditions of this Agreement, in full consideration for the sale, transfer and conveyance of the Shares, at the Closing (i) Buyer shall pay to Seller Ninety-Five Million Dollars (\$95,000,000) (the "<u>Base Cash Portion</u>") and (ii) Buyer shall deliver to Seller One Million Four Hundred Sixty Thousand (1,460,000) shares of the common stock of UEI, par value \$0.01 (the "<u>UEI Shares</u>" or the "<u>Stock Portion</u>" representing

approximately 9.85% of the outstanding share capital of UEI and together with the Base Cash Portion, as finally adjusted, the "Purchase Price"), all payable as follows:

- (a) <u>Payment of Cash Portion</u>. On the Closing Date, Buyer shall pay to Seller Ninety Million Dollars (\$90,000,000) in immediately available funds by bank wire transfer in accordance with the payment instructions and to the account as may be designated by Seller in writing, no later than the fifth (5<sup>th</sup>) business day prior to Closing (the "<u>Seller's Account</u>"). The balance of Five Million Dollars (\$5,000,000) (the "**Holdback Amount**") shall be held back by Buyer. Such holdback constitutes security for the obligations of Seller under Sections 1.3(d)(i) and 1.4(d) of this Agreement and for the indemnification obligations of Seller under Article VII of this Agreement, all in accordance with the provisions and procedures set forth herein, and the Holdback Amount shall be held by Buyer in a separate account (the "**Account**"). Buyer shall release the then-remaining balance of the Holdback Amount (i.e., not being Segregated Funds) in accordance with instructions and to an account designated by Seller at 5:00 p.m., Pacific Time, on the first anniversary of the date hereof (the "**Release Date**") (the period of time from the Closing Date through and including the Release Date is referred to herein as the "**Holdback Period**"), subject to the following:
- (i) Pending the resolution of a Claim or a Third Party Claim (each a "Contested Claim") in accordance with Article VII, upon sending the relevant Claim Notice, Buyer shall set aside and segregate from the Account the lesser of (i) an amount of cash that would be sufficient to completely discharge the Contested Claim in the amount stated in the Claim Notice or (ii) the amount then-remaining in the Account as of the date of the Claim Notice (the "Segregated Funds"); provided, however, only cash that is the subject of a Claim Notice in connection with which Buyer has commenced a litigation proceeding by the serving of a writ during the Holdback Period will constitute Segregated Funds that remain in the Account subsequent to the expiration of the Holdback Period as applicable. The Segregated Funds shall be available for distribution only under the terms and conditions set forth in subsection (ii) and (iii) hereinafter.
- (ii) Buyer shall, within five (5) Business Days of one of the following events (as applicable), release out of the Account and to itself or other applicable Indemnified Party(ies), in connection with a Claim Notice out of the Segregated Funds to the Indemnified Party(ies):
- (A) if Buyer does not receive any Objection to such Claim Notice during such Objection Period, in the amount that is the lesser of (a) the entirety of the Segregated Funds or (b) an amount of Segregated Funds that would be sufficient to completely discharge the amount stated in the Claim Notice; or
- (B) if Buyer receives any Objection to such Claim Notice during the Objection Period, in the amount that is the lesser of (a) the entirety of the Segregated Funds or (b) an amount of Segregated Funds that would be sufficient to completely discharge the amount stated in and in accordance with (x) a written memorandum executed by the applicable parties in accordance with Section 3 hereof and Section 7.6 or (y) a certified copy of a final non appealable order entered against Seller by a court of competent jurisdiction in accordance with

Section 8.1 of the Purchase Agreement, upon receipt from Buyer of a certificate, signed by the corporate secretary of Buyer.

- (iii) The remainder of any amount covered by a Claim Notice and held as Segregated Funds which is not required to be paid to the Indemnified Party(ies) pursuant to subsection (ii) or with respect to which Claim the Buyer Indemnitiees have not taken action to commence legal proceedings within the Holdback Period shall be promptly "de-segregated" and returned to the Account unless such final settlement occurs after the Release Date, in which case such remainder shall be paid to Seller.
  - (b) <u>Delivery of Stock Portion</u>. On the Closing Date, Buyer shall deliver to Seller the Stock Portion.

## 1.3 Net Asset Adjustment.

## (a) Calculation of the Net Assets.

(i) <u>Initial Calculation</u>; <u>Dispute by Seller</u>. As soon as reasonably practicable following the Closing Date, and in any event within twenty (20) Business Days after the date that Enson closes their books with respect to the Closing, Enson shall prepare and deliver to the Auditor for review (i) the closing balance sheet and profit and loss (P&L) statement for Enson, on a consolidated basis, as at the Closing Date, in accordance with the Net Asset Principles and (ii) a calculation of the Net Assets as of the Closing Date based upon such balance sheet and profit and loss statement and adjusted in accordance with the agreed adjustment as set out in <u>Annex C</u>, and the Auditor shall confirm in a report (delivered to Enson, UEI and Seller within forty five (45) calendar days) if there is any adjustments that the Auditor is of the view should be made to the Net Assets and/or the calculation thereof and all documentation in support thereof on the basis of and in accordance with the Net Assets Principles and Annex C and whether there were mathematical errors in the calculation. Enson and Seller shall have ten (10) Business Days from receipt thereof to submit a written notice (a "<u>Dispute Notice</u>") to the other and to Auditor setting forth any dispute that either, in good faith, makes with respect to such report, and Enson, Buyer and Seller shall, in good faith, work to promptly resolve all issues set forth in such Dispute Notice.

## (ii) [Intentionally omitted.]

- (b) <u>Access to Information</u>. Buyer shall promptly make available to the Seller and/or the Auditor, as applicable, all information (as in the possession or control of the Buyer and Enson) and personnel who prepared such information as may be reasonably required to enable the Auditor's preparation of the report pursuant to Section 1.3(a)(i), including, without limitation, the calculation of the Net Assets by Enson and rendering of the Auditor's Decision, as applicable.
  - (c) Costs and Expenses. Enson and Seller shall each bear one-half (1/2) of the Auditor's costs and expenses.
  - (d) Post Closing Purchase Price Adjustment. Within five (5) Business Days of the First Determination Date:

- (i) to the extent that the Net Assets are less than Sixty Eight Million Five Hundred Thousand (\$68,500,000) (the "First Reference Amount"), Seller shall pay to Buyer an amount equal to such difference, plus interest on such amount from the Closing Date at the Prime Rate as of the First Determination Date (computed on the basis of a 365-day year for actual days elapsed), in immediately available funds by (x) joint written instruction of the Buyer and Seller to transfer the difference plus interest from the Account and (y) in case the difference is more than the balance in the Account, by bank wire transfer of such deficit, in accordance with such instructions and to an account designated by Buyer; and
- (ii) to the extent that the Net Assets are more than the First Reference Amount, Buyer shall pay to Seller an amount equal to such difference, plus interest on such amount from the Closing Date at the Prime Rate as of the First Determination Date (computed on the basis of a 365-day year for actual days elapsed), in immediately available funds by bank wire transfer in accordance with such instructions and to an account designated by Seller.

## 1.4 Earnings Adjustment.

- (a) <u>Calculation of Earnings</u>. As soon as reasonably practicable following March 31, 2011, and in any event within ten (10) Business Days after the date that Enson closes their books with respect to March 31, 2011, Buyer shall cause Enson to commission the Auditor to prepare audited consolidated financial statements of Enson for the year ending and as of March 31, 2011 in accordance with the Company Earnings Principles (the "<u>Audited Accounts</u>") and a calculation of the Company Earnings for the year ended March 31, 2011 based upon the Audited Accounts and adjusted in accordance with the agreed adjustment as set out in Annex C within forty five (45) calendar days of confirmation and acknowledgement of its appointment in connection herewith and in any event not later than June 30, 2011. Enson and Seller shall have ten (10) Business Days from receipt thereof to submit a Dispute Notice to the other and to Auditor setting forth any adjustments that either, in good faith, makes to such Audited Accounts and Company Earnings, and Enson, Buyer and Seller shall, in good faith, work to promptly resolve all issues set forth in such Dispute Notice.
- (b) <u>Access to Information</u>. Buyer shall promptly make available to the Seller and the Auditors all information (as in the possession or control of Buyer) and personnel as may be reasonably required to enable the preparation of the Audited Accounts.
  - (c) Costs and Expenses. Enson and Seller shall each bear one-half (1/2) of the Auditor's costs and expenses.
- (d) <u>Post Closing Purchase Price Adjustment</u>. Within five (5) Business Days of the Second Determination Date, to the extent that the Company Earnings are less than Sixteen Million One Hundred And Eighty-Two Thousand (\$16,182,000) (the "<u>Second Reference Amount</u>"), Seller shall pay to Buyer an amount equal to the product of (a) the difference between the Company Earnings and the Second Reference Amount, multiplied by (b) one and one half (1.5), plus interest on such amount from the Closing Date at the Prime Rate as of the Second Determination Date (computed on the basis of a 365-day year for actual days elapsed), in immediately available funds by bank wire transfer in accordance with such instructions and to an account designated by Buyer.

## **ARTICLE II**

## REPRESENTATIONS AND WARRANTIES

- **2.1** Representations and Warranties of Seller. Except as set forth in a disclosure letter (the schedules of exceptions and disclosures) delivered in connection herewith (the "Disclosure Schedules"), Seller hereby represents and warrants to Buyer and UEI as follows:
  - (a) Organization and Standing; Power and Authority; Enforceability.
- (i) Seller is an exempted company with limited liability, duly incorporated, validly existing and in good standing under the laws of the Cayman Islands. Seller has all requisite power and authority (including corporate, as applicable) to enter into the Transaction Documents and perform its obligations hereunder and thereunder. The Transaction Documents have been, or upon execution thereof will be, duly executed and delivered by Seller. The Transaction Documents constitute the valid and binding obligations of Seller, enforceable against Seller in accordance with their respective terms, subject to Laws of general application relating to or affecting the enforcement of creditors' rights generally and to general principles of equity. The Transaction Documents and the transactions contemplated thereby and hereby have been duly approved by all corporate action required by Law, and no other corporate proceedings on the part of Seller or its shareholder(s) are necessary to authorize the Transaction Documents or to consummate the transactions contemplated thereby and hereby.
- (ii) Each of the BVI Companies (a) is a limited liability company duly organized and validly existing under the laws of the British Virgin Islands, (b) except as set forth on Schedule 2.1(a)(ii)(b), only transacts business in the British Virgins Islands and (c) has all requisite corporate power and authority to own, lease and operate its Assets and to carry on the business and operations as it is now being conducted. The BVI Companies do not have any subsidiaries, whether directly or indirectly, (c) do not have any shares of capital stock or any other ownership interest in any Person, and none of the BVI Companies is a partner (general or limited), joint venturer or participant in any partnership, joint venture or other corporate entity except for CG Group's interests in C.G. Timepiece Limited, a company organized under the Laws of the British Virgin Islands ("CG Asia") and Enson's interests in the Subsidiaries.
- (iii) Each of the Hong Kong Companies (a) is a limited liability company duly organized and validly existing under the laws of Hong Kong, (b) only transacts its business in Hong Kong and (c) has all requisite corporate power and authority to own, lease and operate its Assets and to carry on the business and operations as it is now being conducted. None of the Hong Kong Companies has any subsidiaries, whether directly or indirectly, nor owns any shares of capital stock or any other ownership interest in any Person, and none of the Hong Kong Companies is a partner (general or limited), joint venturer or participant in any partnership, joint venture or other corporate entity except for C.G. Development's interests in Gemstar Technology (China) Co., Ltd., a company organized under the Laws of the People's Republic of China ("Gemstar (China)") and Gemstar Technology (Yangzhou) Ltd., a company

organized under the Laws of the People's Republic of China ("Gemstar (Yangzhou)" together with Gemstar (China), the "Gemstar Companies").

- (iv) Each of the Gemstar Companies (a) is a limited liability company duly organized and validly existing under the laws of the People's Republic of China, (b) is duly qualified to transact business as a wholly foreign owned enterprise in the People's Republic of China and (c) has all requisite corporate power and authority to own, lease and operate and to carry on the business and operations as it is now being conducted. Neither of the Gemstar Companies has any subsidiaries, whether directly or indirectly, or owns any shares of capital stock or any ownership interest in any Person, and neither of the Gemstar Companies is a partner (general or limited), joint venturer or participant in any partnership, joint venture or other corporate entity.
- (b) <u>Charter or Constitutional Documents</u>. Copies of the certified charter or constitutional documents of Seller and the Companies delivered to Buyer by Seller pursuant to Section 5.1(g) herein are true, correct and complete in all material respects, and have not been amended or modified since the respective dates thereof.
- (c) <u>Conflicts; Defaults</u>. Neither the execution and delivery of the Transaction Documents, nor the performance by Seller of the transactions contemplated hereby or thereby, will (A) violate, conflict with, or constitute a default under: (1) any of the terms of Seller's or any of the Companies' charter or constitutional documents, (2) any provisions of, or result in the acceleration of any obligation under, any contract, sales commitment, license, purchase order, security agreement, mortgage, note, deed, lien, lease, agreement or instrument, or any order, judgment or decree, relating to the business or operations of Seller or any of the Companies, or by which Seller or any of the Companies is bound or (3) any Law; (B) constitute an event which, after notice or lapse of time or both, would result in such violation, conflict, default, acceleration, or creation or imposition of Liens on any shares or any Assets of any of the Companies; or (C) constitute an event which, after notice or lapse of time or otherwise would create, or cause to be exercisable or enforceable, any option, agreement or right of any kind to purchase any of the shares or any Assets of any of the Companies.

## (d) Capitalization.

- (i) The authorized share capital of Enson is \$80,000.00 divided into 80,000 ordinary shares of nominal value \$1.00 each, of which 60,225 ordinary shares have been issued at nominal value \$1.00 each (the "Shares"). The Shares constitute all issued share capital of Enson. All of the Shares have been duly authorized, are validly issued, fully paid and non assessable and are owned and held of record by Seller.
- (ii) The authorized share capital of C.G. Technology consists of 10,000 non-voting deferred shares of nominal value HK\$1.00 each and 1,000 ordinary shares of nominal value HK\$1.00 each, of which 10,000 non-voting deferred shares and 2 ordinary shares have been issued at nominal value of HK\$1.00 each (the "C.G. Technology Shares"). The C.G. Technology Shares constitute all issued share capital of C.G. Technology. All of the C.G. Technology Shares have been duly authorized, are validly issued, fully paid and non assessable and are owned and held of record by Enson.

- (iii) The authorized share capital of C.G. Development consists of 10,000 ordinary shares of nominal value HK\$1.00, of which 2 ordinary shares have been issued at nominal value of HK\$1.00 (the "C.G. Development Shares"). The C.G. Development Shares constitute all issued share capital of C.G. Development. All of the C.G. Development Shares have been duly authorized, are validly issued, fully paid and non assessable and are owned and held of record by Enson.
- (iv) The entire registered capital of Gemstar (China) is \$8,000,000 has been duly authorized, is validly issued, fully paid and non assessable and is owned and held of record by C.G. Development (the "Gemstar (China) Interests").
- (v) The entire registered capital of Gemstar (Yangzhou) is US\$18,650,000 and has been duly authorized, is validly issued, fully paid and non assessable and is owned and held of record by C.G. Development (the "Gemstar (Yangzhou) Interests").
- (vi) The authorized share capital of Gemstar Polyfirst consists of 10,000 ordinary shares of nominal value HK\$1.00, of which 2 ordinary shares have been issued at nominal value of HK\$1.00 (the "Gemstar Polyfirst Shares"). The Gemstar Polyfirst Shares constitute all issued share capital of Gemstar Polyfirst. All of the Gemstar Polyfirst Shares have been duly authorized, are validly issued, fully paid and non assessable and are owned and held of record: one by Enson and the other one by C.G. Development in trust for Enson.
- (vii) The authorized share capital of CG Group consists of 50,000 ordinary shares of nominal value \$1.00, of which 1 ordinary share has been issued at nominal value of \$1.00 (the "<u>CG Group Share</u>"). The CG Group Share constitutes all issued share capital of CG Group. The CG Group Share has been duly authorized, is validly issued, credited fully paid and non assessable and is owned and held of record by Enson.
- (viii) The authorized share capital of C.G. Timepiece consists of 10,000 ordinary shares of nominal value HK\$1.00, of which 1 ordinary share has been issued at nominal value of HK\$1.00 (the "<u>C.G. Timepiece Share</u>"). The C.G. Timepiece Share constitutes all issued share capital of C.G. Timepiece. The C.G. Timepiece Share has been duly authorized, is validly issued, fully paid and non assessable and is owned and held of record by CG Group.
- (ix) The entire authorized share capital of CG Asia consists of 50,000 ordinary shares of nominal value \$1.00 of which 1 ordinary share has been issued at nominal value of \$1.00 (the "CG Asia Share"). The CG Asia Share constitutes all issued share capital of CG Asia. The CG Asia Share has been duly authorized, is validly issued, fully paid and non assessable and is owned and held of record by CG Group.
- (x) (i) There are no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights, or other contracts or commitments that could require any of the Companies or its shareholder to issue, sell, or otherwise cause to become outstanding any equity interest, (ii) there are no instruments, agreements, arrangements or understandings in respect of the equity of any of the Companies, including without limitation voting trusts, proxies or other similar voting arrangements and (iii)

there are no outstanding or authorized stock appreciation, phantom stock, profit participation or similar rights with respect to any of Companies.

## (e) Title to Shares.

- (i) Seller is the lawful owner, of record and beneficially, of all of the Shares, free and clear of any Liens whatsoever, and with no restrictions on the voting rights and other incidents of record and beneficial ownership pertaining thereto. Except for this Agreement, there are no agreements, contracts or understandings between Seller and any other Person with respect to the acquisition, disposition, transfer or voting of or any other matters pertaining to the Shares. Upon payment in full of the Base Cash Portion and the Stock Portion at Closing pursuant to Section 1.2, good, valid and marketable title to the Shares will pass to Buyer, free and clear of all Liens, and with no restrictions on the voting rights or other incidents of record and beneficial ownership of the Shares.
- (ii) Enson is the lawful owner, of record and beneficially, of all of the Gemstar Polyfirst Shares, C.G. Technology Shares, C.G. Development Shares and CG Group Shares ("<u>Direct Subsidiary Shares</u>"), all free and clear of any Liens whatsoever, and with no restrictions on the voting rights and other incidents of record and beneficial ownership pertaining thereto. Except for this Agreement, there are no agreements, contracts or understandings between Enson and any other Person with respect to the acquisition, disposition, transfer or voting of or any other matters pertaining to the Direct Subsidiary Shares.
- (iii) C.G. Development is, and will remain at Closing, the lawful owner, of record and beneficially, of all of the Gemstar (China) Interests and all of the Gemstar (Yangzhou) Interests, free and clear of any Liens whatsoever, and with no restrictions on the voting rights and other incidents of record and beneficial ownership pertaining thereto. Except for this Agreement, there are no agreements, contracts or understandings between C.G. Development and any other Person with respect to the acquisition, disposition, transfer or voting of or any other matters pertaining to the Gemstar (China) Interests or the Gemstar (Yangzhou) Interests.
- (iv) CG Group is, and will remain at Closing, the lawful owner, of record and beneficially, of all of the C.G. Timepiece Shares and all of the CG Asia Shares, free and clear of any Liens whatsoever, and with no restrictions on the voting rights and other incidents of record and beneficial ownership pertaining thereto. Except for this Agreement, there are no agreements, contracts or understandings between CG Group and any other Person with respect to the acquisition, disposition, transfer or voting of or any other matters pertaining to the C.G. Timepiece Shares or the CG Asia Shares.
- (f) <u>Tangible Property</u>. Each of the Companies owns good, marketable and exclusive title to, or has a valid leasehold in, all tangible personal property, located at the location (in respect of each Company) as set out in <u>Schedule 2.1(f)</u> (including without limitation, all vehicles, machinery, equipment, tools, computer hardware, furniture, fixtures (both real and personal), furnishings and other similar property) necessary for or used in, or held for use in, the conduct of its business and operations as presently conducted (the "<u>Tangible Property</u>"). The Tangible Property is now and at the Closing Date will be free and clear of all Liens of any kind

or nature whatsoever, other than Permitted Liens. The Tangible Property of the Companies, together with the Business Intellectual Property and the Real Property, constitutes all of the material assets, properties, rights and interests necessary in order for Buyer to conduct each of the relevant Company's business and operations as necessary to meet the relevant Company's current contractual obligations. The Tangible Property is generally in good and operating quality, condition and repair, normal wear and tear excepted.

## (g) Real Property.

- (i) Schedule 2.1(g)(i) contains a list of (i) all the owned real property of the Companies (the "Owned Real Property") and (ii) all real property in which the Company has a leasehold interest (the "Leased Real Property"), together with a list of all lease agreements (including all amendments, extensions, renewals, guaranties and other agreements with respect thereto) for each Leased Real Property (each a "Lease"). The Owned Real Property and the Leased Real Property (together, the "Real Property") constitute all of the real property interests which are leased, licensed, used or occupied by the Company.
- (A) With respect to the Owned Real Property, the Company has full title under and subject to the applicable laws to such Owned Real Property free and clear of all Liens, except for Permitted Liens. Gemstar (China) is the registered owner of the Real Property of Gemstar (China) and Gemstar (Yangzhou) is the registered owner of such Real Property of Gemstar (Yangzhou). The land premium or acquisition consideration and other due fees and taxes regarding the acquisition of such Owned Real Property have been fully paid in.
- (B) Each Company has delivered to Buyer a true and complete copy of each Lease. Each Lease constitutes valid and binding obligations of the relevant Company, enforceable against such Company in accordance with its terms, subject to Laws of general application relating to or affecting the enforcement of creditors' rights generally and to general principles of equity.

## (ii) With respect to the Real Property:

- (A) With respect to the Owned Property, no portion thereof is subject to any pending condemnation proceeding by any public or quasi-public authority and there is no threatened condemnation proceeding with respect to any of the Real Property;
- (B) With respect to the Leased Real Property, (i) the rent and all other sums payable under each Lease have been paid to date, (ii) no security deposit or portion thereof deposited with respect to any Lease has been applied in respect of a material breach or default under such Lease which has not been redeposited in full, (iii) all material covenants and conditions contained in each Lease or in any license, consent or other document entered into supplemental to such Lease, on the part of the tenant and the landlord have been observed and performed in all material respects to date and (iv) no breaches have been waived or acquiesced in and there are no rent reviews outstanding or exercisable by any lessor from a date prior to the Closing Date;

- (C) there are no contracts, written or oral, to which any Company is a party, granting to any party or parties the right of use or occupancy of any portion of the parcels of the Real Property;
- (D) none of the Companies has assigned, subleased, mortgaged, deeded in trust or otherwise transferred or encumbered any Real Property or any interest therein and there are no parties (other than Companies) in possession of the Real Property;
- (E) the transactions contemplated hereby do not require the consent of any other party to any Lease; will not result in a breach of or default under any Lease or otherwise cause any Lease to cease to be legal, valid, binding, enforceable and in full force and effect on identical terms following the Closing;
  - (F) there are no outstanding disputes, claims, actions, demands or complaints in respect of any Real Property;
- (G) none of the Companies nor any other party to any Lease is in breach or default under such Lease and there are no existing facts or conditions, which, with the giving of notice, the passage of time or both are reasonably likely to give rise to any such breach or default, or any claim against any Company or any other party to a Lease;
- (H) no event has occurred or circumstance exists which, with the delivery of notice, passage of time or both, would constitute a breach or default under any Lease by any Company or any other party to a Lease, or permit the termination, modification or acceleration of rent under a Lease;
  - (I) there are no forbearance programs in effect with respect to any Lease;
- (J) all of the improvements situated in whole or in part on any Real Property are in operating condition and are usable, adequate and sufficient for the uses to which they are put in the business and operations of the relevant Company, normal wear and tear excepted;
- (K) each Company has obtained or will obtain prior to Closing, any and all consents of the landlord(s) under the Leases and any subleases thereto, and such Lease and subleases shall continue in full force and effect as is in effect as of the date hereof without the payment of any additional sums.
- (h) <u>Material Contracts</u>. The Company has heretofore delivered to Buyer true and correct copies of contracts with Material Suppliers and Material Customers (each, a "<u>Material Contract</u>") which are in full force and effect, in accordance with its terms, subject to Laws of general application relating to or affecting the enforcement of creditors' rights generally and to general principles of equity. Each Company has performed all obligations required to be performed by it to date under the relevant Material Contracts in all material respects and no Company nor any other party to any Material Contract has any material breach or improperly terminated any Material Contract or is in default under any of the Material Contracts in any material respect. No Company has received notice of any default, offset, counterclaim or

defense under any Material Contract and there exists no condition or event which with the giving of notice, the passage of time or both, would reasonably give rise to a material breach, termination or default.

#### (i) Financial Statements.

- (i) Seller has previously delivered to Buyer true and complete copies of (A) the audited consolidated financial statements of Enson for the three fiscal years ended March 31, 2008, March 31, 2009 and March 31, 2010 and unaudited consolidated financial statements of Enson for the five months ended August 31, 2010, (B) the audited financial statements of each of the Hong Kong Companies for the two fiscal years ended March 31, 2008 and March 31, 2009, (C) the unaudited financial statements of each of the Hong Kong Companies for the fiscal year ended March 31, 2010 and five months ended August 31, 2010, (D) the audited financial statements of each of the Gemstar Companies for the three fiscal years ended December 31, 2007, December 31, 2008 and December 31, 2009, (E) the unaudited financial statements of the Gemstar Companies for the three months ended March 31, 2010, for the six months ended June 30, 2010 and for the eight months ended August 31, 2010 (F) the audited financial statements of CG Group for the two fiscal years ended March 31, 2008 and March 31, 2009, (G) the unaudited financial statements of CG Group for the fiscal year ended March 31, 2010 and five months ended August 31, 2010, (H) the audited financial statements of CG Asia for the two fiscal year ended March 31, 2008 and March 31, 2009, (I) the unaudited financial statements of CG Asia for the fiscal year ended March 31, 2010 ((A) through (I) collectively, the "Financial Statements").
- (ii) Each of the Financial Statements (i) is true, complete and correct in all material respects, (ii) was prepared from the accounting Books and Records kept by the relevant Company for its business and operations (which accounting Books and Records are true, correct and complete in all material respects), in accordance with HKFRS and generally accepted accounting principles in the People's Republic of China (as the case may be) applied on a consistent basis and (iii) presents a true and fair view in all material respects the financial position and results of operations of the relevant Company as of and for the periods then ended (subject to normal recurring year-end adjustments in the case of the unaudited financial statements and the absence of footnote disclosure), consistent with state and local Tax Returns filed by the relevant Company save and except depreciation allowance and tax adjustments.
- (j) <u>Absence of Undisclosed Liabilities</u>. No Company has any material debts, liabilities or obligations of any nature relating to or in connection with its business or operations that are not (i) set forth on the face of the unaudited consolidated financial statements and disclosure notes of Enson as at August 31, 2010 or (ii) other than through any breach or default by the Companies, incurred in the ordinary course of business consistent with past practice since August 31, 2010.
- (k) <u>Accounts Receivable</u>. All accounts receivable outstanding as at the Closing Date will represent sales actually made in the ordinary course of business consistent with past practice and are subject to no counterclaims, setoffs or rights to return reasonably likely to interfere with full and timely collection of, any of such outstanding accounts receivable other than routine credits granted for errors in ordering, shipping, pricing, defects or similar matters.

Seller has previously delivered to Buyer an aged listing by customer of the accounts receivable that were outstanding as of September 30, 2010, which is true and correct in all material respects.

- (l) <u>Litigation</u>. No Company is subject to any order of, or written agreement or memorandum of understanding with, any Governmental Authority, and there exists no litigation, action, suit, claim, audit or proceeding pending or threatened against or affecting any Company or its business, operations or Assets, at law or in equity or before any Governmental Authority, and no one has valid grounds to assert any such litigation, action, suit, claim or proceeding. There are no actions, suits, claims, audits or proceedings pending or threatened, seeking to or that would reasonably be expected to prevent, hinder, modify, delay or challenge the transactions contemplated by this Agreement. The Companies have provided to the Buyer a list description of (A) all litigation, actions, suits, investigations, claims, audits and proceedings asserted, brought or threatened against each of the Companies since January 1, 2007, together with a description of the outcome or present status thereof and (B) all judgments, orders, decrees, writs or injunctions entered into by or against each of the Companies that remain in effect or have not yet been complied with by each Company. There is no action, suit, claim, audit or proceeding brought by any Company against any Person that is pending.
- (m) <u>Customers and Suppliers</u>. No Material Customer or Material Supplier has terminated or, to the Knowledge of Seller, intends to terminate its relationship with such Company, has decreased or limited or intends to decrease or limit its purchases or supplies (as applicable) or would not continue to purchase or supply from or to the Company, nor has any such Material Customer or Material Supplier indicated to the Company that it intends to do so.
- (n) <u>Compliance with Law; Illegal Payments</u>. Each of the Companies has maintained, and has caused its directors, officers, agents, employees or any other Persons acting on its or their behalf to maintain, its Assets and operated its business and operations in compliance with all applicable laws and regulations including anti-corruption or anti-bribery laws and regulations, in all material respects.
- (o) <u>Brokers, Finders and Agents</u>. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the transactions contemplated by this Agreement or any Transaction Document based upon arrangements made by or on behalf of a Company.
- (p) Intellectual Property. Collectively, the Companies own and possess, free and clear of all Liens, all right, title and interest in and to, or have the rights to use, all Intellectual Property necessary or desirable for the operation of the business and operations of each of the Companies, as applicable, as currently conducted (the "Business Intellectual Property."). No Company has infringed, misappropriated or otherwise conflicted with or violated, and neither the operation of the business and operations of any of the Companies nor the ownership or use of any of the Business Intellectual Property infringes, misappropriates or otherwise conflicts with or violates, any Intellectual Property rights of any third party. To the Knowledge of Seller, no third party has infringed, misappropriated or otherwise conflicted with or violated any of the Business Intellectual Property. All of the Business Intellectual Property shall be owned or available for use by the Companies immediately after the Closing.

- (q) <u>Permits</u>. Each of the Companies is in full compliance in all material respects with each Permit set forth on <u>Schedule 2.1(q)</u>, as applicable. Such Permits constitute all necessary Permits that are required by Law or are otherwise necessary for the operation of the business and operations of each Company as the same are currently being operated and for ownership or use of its Assets, as applicable. All such Permits are in full force and effect and will be in full force and effect as at the Closing Date. There has been no material change in the facts or circumstances reported or assumed in the application for or granting of such Permits that would result in the invalidity or revocation of any Permit. None of the Companies has received written notice from any Governmental Authority regarding any proposed modification, non-renewal, suspension, revocation or cancellation of any Permits, and no event has occurred which could reasonably be expected to result in the modification, non-renewal, suspension or cancellation of any such Permits.
- (r) <u>Labor Matters</u>. Seller has previously delivered to Buyer a form of the standard employment contracts and non-competition contracts to which each of the Companies is a party or by which each of the Companies is bound. No Company is a party to any collective bargaining agreement or other labor union contract. No collective bargaining agreement is being negotiated by any Company. No Company has agreed to bargain with any labor organization. No Company has agreed to recognize any union or other collective bargaining unit. No petition for representation has been filed, and no union has been certified as representing the employees of the Company. No Company has been fined or ordered to pay any overdue payment by any competent authority due to its default or delay in the payment of the due salary or social security fees. To the Knowledge of Seller, there are no activities or proceedings of any labor union employees, or other party(ies), to organize the employees of any Company.
- (s) <u>Employee Plans</u>. No Company has any employee benefit plans (each an "<u>Employee Plan</u>" and collectively, "<u>Employee Plans</u>") in effect on the date of this Agreement, and no Company has had any Employee Plans in effect for the five (5) years preceding the date of this Agreement.

## (t) Environmental Matters.

- (i) There are no underground storage tanks on any Owned Real Property or the Leased Real Property.
- (ii) Seller has previously provided to Buyer true and complete copies of all material reports of environmental investigations from the three years prior to the date hereof, including, but not limited to, environmental site assessments, conducted on any portion of any property currently used by the Company in its business or any property that is in any Company's possession.
- (u) <u>Absence of Certain Changes in Circumstances</u>. Since August 31, 2010, there has not occurred any material adverse change in the financial condition, results of operations, Assets or Liabilities of each Company and each Company's business and operations has been conducted in the ordinary course of business consistent with past practice.

## (v) Taxes.

- (i) Each of the Companies has duly prepared and filed or caused to be prepared and filed all Tax Returns required to be filed by it with any Governmental Authority. Each such Tax Return is true, correct and complete in all material respects. All Taxes owed to any Governmental Authority by any Company for periods covered by such Tax Returns and all claims, demands, assessments, judgments, costs and expenses connected therewith, have been paid in full on or before the relevant due date.
- (ii) No Hong Kong Company or BVI Company is required to withhold any Taxes and each of the Gemstar Companies has withheld and paid over all Taxes required to have been withheld and paid over, and complied with all information reporting and backup withholding requirements, including maintenance of required records with respect thereto, in connection with amounts paid or owing to any employee, creditor, independent contractor, or other third party. No Company has any Liability for Taxes of any Person, as a transferor, transferee or successor, by contract, or otherwise. There are no Liens or other encumbrances with respect to Taxes upon any of the Assets other than with respect to Taxes not yet due and payable.
- (iii) No Company has any Tax deficiency proposed or assessed against it, nor has any Company executed any waiver of any statute of limitations on the assessment or collection of any Tax or governmental charge. No Company is a party to any action or proceeding nor is any such action or proceeding threatened, for the assessment or collection of any Taxes and no deficiency notices or reports have been received by any Company in respect of any Tax. There are no outstanding waivers or agreements extending the statute of limitations for any period with respect to any Tax to which the Company may be subject.
  - (iv) No Company has ever been a member of an affiliated group filing consolidated or combined Tax Returns since January 1, 2000.
- (v) Each of the Companies has furnished to Buyer correct and complete copies of all Tax Returns, examination reports, and statements of deficiencies assessed against or agreed to by each of the Companies filed or received since its taxable year ending March 31, 2009 (for the Hong Kong Companies and the BVI Companies), and December 31, 2009 (for Gemstar Companies). No claim has been made by an authority in a jurisdiction where any Company does not file Tax Returns that it is or may be subject to taxation by that jurisdiction.
- (vi) To the best knowledge of the Seller, each of the Companies will remain entitled to all the tax incentives and other tax preferential treatments it is currently entitled to on term and conditions according to the relevant tax law and regulations.
- (w) <u>Bank Accounts</u>. The Seller has delivered to the Buyer a schedule setting out the names and locations of all banks, trust companies, savings and loan associations and other financial institutions at which any of the Companies maintains any safe deposit boxes or accounts (specifying the identifying numbers), and the names of all Persons authorized to draw thereon, make withdrawals therefrom or have access thereto.

- (x) <u>Books and Records</u>. The Books and Records of each of the Companies is a materially accurate record of transactions relating to each of the Companies in all material respects, and have been maintained on a consistent practice in all material respects.
  - 2.2 Stock Portion Related Representations and Warranties. Seller hereby represents and warrants to Buyer as follows:
    - (a) Seller recognizes that no Governmental Authority has recommended or endorsed the purchase of the UEI Shares.
- (b) Seller is acquiring the UEI Shares for its own account as principal, not, in whole or in part, as a nominee or agent for or otherwise on behalf of any other Person, for investment purposes only, and not with a view to, or for, resale, distribution or fractionalization thereof.
- (c) Seller has not been furnished with any oral representation or information in connection with the offering of the UEI Shares that is not contained in, or is in any way contrary to or inconsistent with, statements made herein or in documents regarding UEI available at www.sec.gov.
- (d) Seller is not subscribing for the UEI Shares as a result of or subsequent to any advertisement, article, notice or other communication published in any newspaper, magazine or similar media or broadcast over the Internet, television or radio or presented at any seminar or meeting or any public announcement or filing of or by UEI.
- (e) No representations or warranties have been made to Seller by Buyer or UEI, or any officer, employee, agent, affiliate or subsidiary of Buyer or UEI, other than the representations of Buyer and UEI contained herein, and in subscribing for the UEI Shares, Seller is not relying upon any representations other than those contained herein.
- (f) Seller understands and acknowledges that UEI does not intend to register the UEI Shares under the Securities Act, or to assist the undersigned in complying with the securities laws of any state of the United States or of any foreign jurisdiction.
- (g) Seller understands and acknowledges that the UEI Shares are being offered and sold in reliance on an exemption from the registration requirements of United States federal and state securities laws under Regulation S promulgated under the Securities Act and that UEI is relying in part upon the truth and accuracy of the representations, warranties, agreements, acknowledgments and understandings of Seller set forth herein in order to determine the applicability of such exemptions and the suitability of Seller to acquire the UEI Shares. In this regard, Seller further represents, warrants and agrees that:
- (i) Seller is not a U.S. Person (as defined below) and is not an affiliate (as defined in Rule 501(b) under the Securities Act) of Buyer or UEI and is not acquiring the UEI Shares for the account or benefit of a U.S. Person. A U.S. Person means any one of the following:
  - (A) any natural person resident in the United States of America;

- (B) any partnership or corporation organized or incorporated under the laws of the United States of America;
- (C) any estate of which any executor or administrator is a U.S. person;
- (D) any trust of which any trustee is a U.S. person;
- (E) any agency or branch of a foreign entity located in the United States of America;
- (F) any non-discretionary account or similar account (other than an estate or trust) held by a dealer or other fiduciary for the benefit or account of a U.S. person;
- (G) any discretionary account or similar account (other than an estate or trust) held by a dealer or other fiduciary organized, incorporated or (if an individual) resident in the United States of America; and
- (H) any partnership or corporation if: (A) organized or incorporated under the laws of any foreign jurisdiction; and (B) formed by a U.S. person principally for the purpose of investing in securities not registered under the Securities Act, unless it is organized or incorporated, and owned, by accredited investors (as defined in Rule 501(a) under the Securities Act) who are not natural persons, estates or trusts.
- (ii) Any resale of the UEI Shares during the 'distribution compliance period' as defined in Rule 902(f) to Regulation S shall only be made in compliance with exemptions from registration afforded by Regulation S; pursuant to registration of the UEI Shares under the Securities Act; or pursuant to an available exemption from the registration requirements of the Securities Act, and Seller agrees not to engage in hedging transactions with regard to the UEI Shares unless in compliance with the Securities Act;
- (iii) At the time of the origination of contact concerning this Agreement and the date of the execution and delivery of this Agreement, Seller was outside of the United States.
- (iv) Neither the Seller nor or any person acting on its behalf has engaged, nor will engage, in any directed selling efforts to a U.S. Person with respect to the UEI Shares and the Seller and any person acting on his behalf have complied and will comply with the "offering restrictions" requirements of Regulation S under the Securities Act.
- (v) Seller acknowledges that the UEI Shares will be considered "restricted securities" within the meaning of Rule 144 under the Securities Act notwithstanding any resale transactions made in compliance with Rules 901 and 904 under Regulation S.
- (vi) Seller understands that any and all certificates representing the UEI Shares, if and when certificated, and any and all securities issued in replacement thereof or in

exchange therefor shall bear the following legends or ones substantially similar thereto, which Seller has read and understands:

- (A) "THE SECURITIES ARE BEING OFFERED AND SOLD TO INVESTORS WHO ARE NOT U.S. PERSONS (AS DEFINED IN REGULATION S UNDER THE SECURITIES ACT OF 1933, AS AMENDED ("THE SECURITIES ACT")) AND WITHOUT REGISTRATION WITH THE UNITED STATES SECURITIES AND EXCHANGE COMMISSION UNDER THE SECURITIES ACT IN RELIANCE UPON REGULATION S PROMULGATED UNDER THE SECURITIES ACT."
- (B) "TRANSFER OF THESE SECURITIES IS PROHIBITED, EXCEPT IN ACCORDANCE WITH THE PROVISIONS OF REGULATION S, PURSUANT TO REGISTRATION UNDER THE SECURITIES ACT, OR PURSUANT TO AVAILABLE EXEMPTION FROM REGISTRATION. HEDGING TRANSACTIONS MAY NOT BE CONDUCTED UNLESS IN COMPLIANCE WITH THE SECURITIES ACT."
- (vii) Seller consents to UEI making a notation on its records or giving instructions to any transfer agent of UEI in order to implement the restrictions on transfer of the UEI Shares set forth in this Section 2.2.
- **2.3** Representations and Warranties of Buyer. Except as set forth on the Disclosure Schedules, Buyer hereby represents and warrants to Seller as follows:
- (a) <u>Organization and Standing</u>; <u>Corporate Power and Authority</u>. Buyer is a company duly organized, validly existing and in good standing under the laws of Hong Kong. Buyer has all requisite power and authority (including corporate) to enter into the Transaction Documents and perform its obligations hereunder and thereunder. The Transaction Documents have been, or upon execution thereof will be, duly executed and delivered by Buyer. The Transaction Documents constitute the valid and binding obligations of Buyer, enforceable against Buyer in accordance with their respective terms, subject to Laws of general application relating to or affecting the enforcement of creditors' rights generally and to general principles of equity. The Transaction Documents and the transactions contemplated thereby and hereby have been duly approved by all corporate action required by Law, and no other corporate proceedings on the part of Buyer or its shareholder(s) are necessary to authorize the Transaction Documents or to consummate the transactions contemplated thereby and hereby.
- (b) <u>Conflicts; Defaults</u>. Neither the execution and delivery of the Transaction Documents, nor the performance by Buyer of the transactions contemplated hereby or thereby, will violate, conflict with, or constitute a default under: (i) any of the terms of Buyer's certificate of incorporation or bylaws, (ii) any provisions of, or result in the acceleration of any obligation under, any contract, sales commitment, license, purchase order, security agreement, mortgage, note, deed, lien, lease, agreement or instrument, or any order, judgment or decree, relating to the business or operations of Buyer, or by which Buyer is bound or (iii) any Law.
- (c) <u>Brokers, Finders and Agents</u>. Buyer is not directly or indirectly obligated to anyone as a broker, finder or investment banker in any other similar capacity in connection with this Agreement or the transactions contemplated hereby.

## 2.4 Representations and Warranties of UEI. UEI hereby represents and warrants to Seller as follows:

- (a) Buyer has obtained a binding commitment for the provision of debt financing in an amount which, together with other funds available to Buyer, is sufficient to fund the transactions contemplated by this Agreement in accordance with the terms hereof.
- (b) The UEI Shares have been duly authorized and, when issued and paid for, will be validly issued, fully paid and non-assessable. The UEI Shares are not and will not be subject to the preemptive rights of any holders of any security of Buyer and all corporate action required to be taken for the authorization, issuance and sale of the UEI Shares has been duly and validly taken.
- (c) Except as disclosed in UEI's Annual Report on Form 10-K for the fiscal year ended December 31, 2009 and all reports required to be filed by UEI under the '34 Act, including pursuant to Section 13(a) or 15(d) thereof, since the date of such filing (the foregoing materials, all exhibits thereto, and all documents, reports and information incorporated therein by reference are collectively referred to herein as the "Commission Reports"), there are no legal or governmental actions, suits or proceedings pending or threatened to which UEI or any of its subsidiaries is or may be a party or of which property owned or leased by UEI or any of its subsidiaries is or may be the subject, or related to environmental or discrimination matters, which actions, suits or proceedings, would reasonably be expected, individually or in the aggregate to have a material adverse effect on UEI's business, taken as a whole. Neither UEI nor any of its subsidiaries is a party or subject to the provisions of any injunction, judgment, decree or order of any court, regulatory body, administrative agency or other governmental body which would reasonably be expected, individually or in the aggregate, to have a material adverse effect on UEI's business, taken as a whole.
- (d) UEI has filed all Commission Reports, for the two years preceding the date hereof on a timely basis or has received a valid extension of such time of filing and has filed any such Commission Reports prior to the expiration of any such extension. As of their respective dates, the Commission Reports complied in all material respects with the requirements of the '34 Act and the rules and regulations thereunder, and none of the Commission Reports, when filed, contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.
- (e) All agreements required to be filed as exhibits to all reports required to be filed by UEI under the '34 Act" since the UEI's most recent Annual Report on Form 10-K under Item 601 of Regulation S-K to which UEI or any of its subsidiaries is a party, have been filed by UEI as exhibits to such reports (the "<u>UEI Material Contracts</u>"). The UEI Material Contracts have been duly authorized, executed and delivered by UEI or its subsidiaries, constitute valid and binding agreements of UEI or its subsidiaries (as applicable) and are enforceable against UEI or its subsidiaries (as applicable) in accordance with their respective terms.
- (f) UEI has not, in the twelve (12) months preceding the date hereof, received notice from the Nasdaq Stock Market (the "Principal Market") to the effect that UEI is not in

compliance with the listing or maintenance requirements of the Principal Market. Except as described in the Commission Reports, UEI has no reason to believe that it will not in the foreseeable future continue to be in compliance or regain compliance in a timely manner, as the case may be, with all such listing and maintenance requirements. The issuance of the UEI Shares hereunder does not contravene the rules and regulations of the Principal Market. The Common Stock has been registered pursuant to Section 12(b) of the Exchange Act and is currently listed on the Principal Market.

- (g) The financial statements of UEI, together with the related schedules and the notes thereto, included in the Commission Reports comply in all material respects with applicable accounting requirements and the applicable requirements of the '34 Act as in effect at the time of filing. Such financial statements have been prepared in accordance with generally accepted accounting principles applied on a consistent basis during the periods involved ("GAAP"), except as may be otherwise specified in such financial statements or the notes thereto, and fairly present in all material respects the financial position of UEI as of and for the dates thereof and the results of operations and cash flows for the periods then ended, subject, in the case of unaudited statements, to normal, immaterial, year-end audit adjustments. Grant Thornton LLP, who has audited certain financial statements of UEI, have been appointed by UEI's audit committee or by the Board of Directors, as the case may be.
- (h) UEI is in compliance in all material respects with all provisions of the Sarbanes-Oxley Act of 2002, as amended, applicable to it, and the applicable rules and regulations promulgated thereunder by all government and regulatory authorities and agencies. UEI maintains a system of internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with generally accepted accounting principles in the United States and to maintain accountability for assets, (iii) access to assets is permitted only in accordance with management's general or specific authorization and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect thereto. UEI has established and maintains and evaluates "disclosure controls and procedures" (as such term is defined in Rule 13a-15 and Rule 15d-15 under the '34 Act) and "internal control over financial reporting" (as such term is defined in Rule 13a-15 and Rule 15d-15 under the '34 Act). UEI's certifying officers have evaluated the effectiveness of UEI's disclosure controls and procedures and UEI presented in its Form 10-K for the most recently ended fiscal year the conclusions of UEI's certifying officers about the effectiveness of such disclosure controls and procedures.
- (i) None of UEI, its subsidiaries, any of their affiliates, or any Person acting on their behalf has, directly or indirectly, made any offers or sales of any security or solicited any offers to buy any security, under circumstances that would require the sale of any of the Securities pursuant to this Agreement to be registered under the '33 Act, whether through integration with prior offerings or otherwise. None of the UEI, its subsidiaries, their Affiliates or any Person acting on their behalf will take any action or steps referred to in the preceding sentence that would (i) require the sale of any of the Securities pursuant to this Agreement to be registered under the '33 Act, except to the extent that the costs and obligations thereof are not borne by Seller, (ii) cause the sale of the UEI Shares pursuant to this Agreement to be integrated

with other offerings in violation of the '33 Act or (iii) cause the issuance of UEI Shares to be subject to any stockholder approval requirement.

**2.5** Further Assurances by Buyer and UEI. Buyer and UEI shall use all best efforts to implement the provisions of this Agreement, and for such purpose, Buyer and UEI, at the request of Seller, at or after the Closing, shall, without further consideration, promptly execute and deliver, or cause to be executed and delivered, to Seller such deeds, assignments, bills of sale, consents and other instruments and documents in addition to those required by this Agreement, in form and substance satisfactory to Seller, and take all such other actions, as Seller may reasonably deem necessary to (i) implement any provision of this Agreement, (ii) to more effectively transfer, convey and assign to Seller and its shareholders good and marketable title to, and to put Seller in actual possession and control of, all of the UEI Shares, free and clear of all Liens.

# ARTICLE III COVENANTS

- **3.1 Closing.** Closing shall occur simultaneously with the signing of this Agreement.
- **3.2** <u>Confidentiality.</u> Seller shall, and shall cause its Affiliates, counsel, accountants, advisors, directors, officers, then-current employees, investment bankers, agents or other representatives, as applicable, to, hold in strict confidence and not use any information which remains after Closing in the possession of Seller or its Affiliates or their respective counsel, accountants, advisors, directors, officers, then-current employees, investment bankers, agents or other representatives, as applicable, concerning the Companies, their businesses and operations, the Assets or the terms and provisions of this Agreement and the transactions contemplated hereby. Seller shall not release or disclose any such information to any Person other than Buyer.
- **3.3** Further Assurances. Seller shall use all best efforts to implement the provisions of this Agreement, and for such purpose, Seller, at the request of Buyer, at or after the Closing, shall, without further consideration, promptly execute and deliver, or cause to be executed and delivered, to Buyer such deeds, assignments, bills of sale, Consents and other instruments and documents in addition to those required by this Agreement, in form and substance satisfactory to Buyer, and take all such other actions, as Buyer may reasonably deem necessary to (i) implement any provision of this Agreement, (ii) to more effectively transfer, convey and assign to Buyer good and marketable title to, and to put Buyer in actual possession and operating control of, all of the Shares, free and clear of all Liens and (iii) allow Buyer to conduct the business and operations of the Companies, as applicable, in the ordinary course of business consistent with past practice from and after the Closing Date, including without limitation cooperating and coordinating with Buyer in connection with any litigation, action, suit, investigation, claim or proceeding challenging the legality of, or seeking to investigate, restrain, prohibit or materially modify, the transactions contemplated in this Agreement instituted by the FTC, the Antitrust Division of the DOJ or other Governmental Authorities under any applicable foreign antitrust of competition Laws.

- **3.4** <u>Maintenance of Existence</u>; <u>Adequate Funds</u>. Seller shall preserve, renew and keep in full force and effect its existence, and shall not take any actions, including paying any dividends, that would render or is reasonably likely to render Seller Insolvent, or unable to pay and discharge its obligations hereunder, including those pursuant to ARTICLE VII hereof.
- **3.5** <u>Consents.</u> Seller has caused and shall continue to cause the Companies to commence and use all best efforts to obtain or satisfy, promptly after the Closing at the earliest practicable date, all Consents set forth on <u>Schedule 3.5</u>. If any such Consents are not obtained and/or satisfied by the time of the Closing and Buyer in its sole discretion waives such requirement, then, at Seller's sole expense, Seller shall continue to use all such best efforts to obtain the same.
- **3.6** <u>Transfer and Reorganization Taxes</u>. While the Shares are not Hong Kong Stock (as defined in the Stamp Duty Ordinance (Cap. 117 of the Laws of Hong Kong), in the event that Hong Kong stamp duty shall be payable on the contract notes and the instrument of transfers in respect of the transfer of the Shares as contemplated by this Agreement ("<u>Transfer Taxes</u>"), it shall be born equally by Seller, on the one hand, and Buyer, on the other. Each Party shall file all necessary documentation in connection with the payment and reporting of Transfer Taxes.

## **ARTICLE IV**

## **CERTAIN ADDITIONAL COVENANTS**

- **4.1** Expenses. Seller, on the one hand, and Buyer and UEI on the other hand, shall each bear their own legal, accounting, and other expenses (the "Expenses") in connection with the negotiation, preparation and execution of the Transaction Documents and the transactions contemplated hereby and thereby.
- **4.2** Press Releases and Disclosure. Save in respect of the notification by the Seller to the relevant Hong Kong regulatory authority of the entering into and closing of the transactions contemplated hereunder, no Party shall, and each Party shall cause its Affiliates, counsel, accountants, advisors, directors, officers, employees, investment bankers, agents and other representatives not to, issue or cause publication of any press release or other announcement or public communication with respect to this Agreement or the transactions contemplated hereby or otherwise disclose this Agreement or the transactions contemplated hereby to any third party without the consent of the other Parties hereto, which consent shall not be unreasonably withheld except (a) as required by law court order or judicial process and after conferring with the other parties concerning the timing and content of such required disclosure, and (b) in the case of Buyer or UEI, (i) as may be required of Buyer or UEI by, or as may be advisable under, applicable Securities and Exchange Commission laws, rules or regulations, by any, or as required by a stock exchange. Buyer and UEI shall consult in good faith with Seller prior to issuing a press release concerning this Agreement and the transactions contemplated hereby, but Seller's consent is not required prior to issuing such press release.
- **4.3** Government and other Regulatory Approvals. Each Party shall, and shall cause its appropriate respective Affiliates to, use all commercially reasonable efforts to obtain

any authorizations, consents, novations, orders and approvals of, with or by any Governmental Authority for such Party necessary for the performance by such Party of its obligations under the Transaction Documents, and the consummation of the transactions contemplated hereby and thereby, including without limitation, in connection with any foreign antitrust or competition Laws and shall cooperate fully with each other in all reasonable respects in promptly seeking to obtain such authorizations, consents, orders and approvals. Neither Seller nor Buyer shall take any action that will have the effect of delaying, impairing or impeding the receipt of any required regulatory approvals.

- **4.4** Reporting Requirements and Furnishing of Information. As long as Seller and its shareholders owns the UEI Shares, UEI shall remain a "reporting issuer" as defined in Rule 902(i) to Regulation S, subject to ten (10) calendar days written notice otherwise by UEI to Seller. UEI further covenants that it will take such further commercially reasonable action as Seller may reasonably request, all to the extent required from time to time to enable Seller to sell the Securities without registration under the '33 Act within the limitation of the exemptions provided by Rule 144.
- **4.5 Piggy-back Registrations.** If at any time during which Seller holds the UEI Shares, there is not an effective registration statement covering the resale of UEI Shares by Seller or its permitted transferees, and UEI shall determine to prepare and file with the Commission a registration statement relating to an offering for its own account or the account of others under the '33 Act of any of its equity securities, other than on Form S-4 or Form S-8 (each as promulgated under the '33 Act) or their then equivalents relating to equity securities to be issued solely in connection with any acquisition of any entity or business or its shares of common stock issuable in connection with stock option or other employee benefit plans, then UEI shall promptly send to Seller written notice of such determination and, if within fifteen days after receipt of such notice, Seller or its permitted transferees holding, in the aggregate, at least twenty (20%) percent of the UEI Shares issued as the Stock Portion hereunder shall so request in writing, UEI shall include in such registration statement all or any part of such UEI Shares not theretofore registered Seller requests to be registered, subject to customary underwriter cutbacks after any other holders of registration rights have been cutback in full. All expenses incurred in connection with any such registration of the UEI Shares shall be borne solely by UEI; provided that Seller or other transferee shall be solely responsible for any commissions, underwriting discounts or brokerage fees in connection with the sale of UEI Shares thereunder. Seller shall have the right to assign its rights under this Section 4.5 to any transferee of the UEI Shares, so long as the transfer is in compliance with an applicable exemption from the registration requirements of the '33 Act. In connection with any such registration, Seller or any permitted transferee shall be required to provide selling shareholder questionnaires and other customary documents as a condition to inclusion of such UEI Shares in the registration statement. Nothing contained herein shall prohibit UEI from determining, at any time, not to file a registration statement or, if filed, to withdraw such registration or terminate or abandon the offering related thereto. Notwithstanding anything in this Section 4.5 to the contrary, UEI shall not be obligated to include any UEI Shares issued hereunder in any registration statement in the event that the holder thereof may then sell all of such holder's UEI Shares issued hereunder to the public in accordance with Rule 144 under the '33 Act or any successor provision thereof, without regard to any volume restrictions contained in such Rule 144.

- **4.6** <u>Post-Closing Operation of the Company.</u> Subject to the terms of this Agreement, subsequent to the Closing, Buyer shall have sole discretion with regard to all matters relating to the operation of the Company; <u>provided</u>, that through March 31, 2011, Buyer shall not, directly or indirectly, take any actions in bad faith that would have the purpose of increasing any amounts which may be payable to Buyer pursuant to Section 1.4(c) hereunder and shall allow Chan Tak Chung, Sin Kwong Sang and Ho Yin Ping to conduct the Company's daily operation in a manner materially consistent with its past practice.
- **4.7** Transfer of Shares to shareholders of Seller. UEI hereby acknowledges that the Seller intends to distribute the UEI Shares to its shareholders relying on exemption under Reg S, and UEI shall instruct its transfer agent to record the issuance of the UEI Shares accordingly.
- **4.8** Removal of Restricted Legend. Upon expiration of the six month restriction period under Regulation S and Rule 144, UEI shall instruct its transfer agent to remove the restrictions related to the '33 Act from the UEI Shares and to re-issue such shares free of any restrictions under the '33 Act, to the extent Seller has submitted a written notice to UEI identifying the name of each holder of such shares and the number of such shares owned by each such holder and to the extent UEI has determined from such written notice that such shares are eligible to have such restrictions removed (such determination not to be unreasonably withheld, conditioned or delayed).

#### **ARTICLE V**

#### **CLOSING DELIVERIES**

- 5.1 Seller's Deliveries. At the Closing, Seller shall deliver or cause to be delivered to Buyer (any of which may be waived exclusively by Buyer):
- (a) Original certificates evidencing ownership of the Shares and other documents to transfer title to the Shares to Buyer and instruments of transfer duly executed by Seller in favor of Buyer relating to transfer of the Shares.
  - (b) A certificate signed by Seller to the effect that the following conditions have been fulfilled:
- (i) <u>Representations and Warranties</u>. Each of the representations and warranties of Seller made in Section 2 of this Agreement, (i) if specifically qualified by materiality, shall be true and correct in all respects, in each case as of the Closing Date as though made at such time (except where any such representation or warranty is specific as of an earlier date, in which event it shall remain true and correct (as qualified, if applicable) as of such earlier date).
- (ii) <u>Covenants</u>. Seller shall have performed and complied in all respects with all covenants, obligations and agreements required to be performed or complied with by it on or prior to the Closing Date.

- (iii) No Proceeding or Litigation. No litigation, action, suit, investigation, claim or proceeding challenging the legality of, or seeking to investigate, restrain, prohibit or materially modify, the transactions contemplated in this Agreement shall have been instituted and not settled or otherwise terminated, including without limitation under any foreign antitrust or competition Laws.
- (c) An incumbency certificate by each officer or director of Seller who will execute a document in connection with the transactions contemplated hereby on behalf of Seller in the form set out in Exhibit D hereto.
- (d) (a) Certified copy of the resolution of the Board of Directors of Seller, authorizing and approving the transactions contemplated hereby and the execution and delivery of the Transaction Documents; (b) certified copy of the resolution of the shareholders of Seller, authorizing and approving the transactions contemplated hereby and the execution and delivery of the Transaction Documents; and (c) copy of Seller's charter or constitutional documents, certified by a director of Seller, respectively, to be true, correct, complete and in full force and effect and unmodified as of the Closing Date.
  - (e) Resolutions of the board of directors of each Company approving the transfer of the Shares by Seller to Buyer and the change of directors.
- (f) Such other deeds, bills of sale, endorsements, assignments, affidavits, and other good and sufficient instruments of sale, assignment, conveyance and transfer in form and substance satisfactory to Buyer and its counsel, as are required to effectively vest in Buyer good and marketable title in and to all of the Shares, free and clear of any and all Liens, and to allow Buyer to conduct the business and operations of the Companies in the ordinary course of business consistent with past practice from and after the Closing Date.
- (g) The Companies' charter or constitutional documents, certified by a director or the corporate secretary of each of the Companies, and the legal representative of the Gemstar Companies to be true, correct, complete and in full force and effect and unmodified as of the Closing Date.
- (h) The Non Competition Agreement in the form attached hereto at Exhibit B (the "Non Competition Agreement") executed by Seller and Lau Luen Hung Thomas.

#### (i) [Intentionally omitted.]

- (j) A Key Employee Employment Agreement, in the form attached hereto as Exhibit C, executed by each of Chan Tak Chung, Sin Kwong Sang and Ho Yin Ping.
  - (k) Letters effecting the resignation from the Board of Directors of each Company from each director thereof.
  - (l) The Books and Records.
  - (m) The Disclosure Schedules.

- **5.2 Buyer's Deliveries.** At the Closing, Buyer shall deliver or cause to be delivered to Seller (any of which may be waived exclusively by Seller):
- (a) Confirmation from the transfer agent of UEI that the transfer agent has recorded the issuance of the UEI Shares to Seller.
- (b) A certificate signed by Buyer to the effect that the following conditions have been fulfilled:
- (i) <u>Representations and Warranties</u>. Each of the representations and warranties of Buyer made in Sections 2.3 and 2.4 of this Agreement (i) if specifically qualified by materiality, shall be true and complete as so qualified and (ii) if not qualified by materiality, shall be true and correct in all material respects, in each case as of the Closing Date as though made at such time (except where any such representation or warranty is specific as of an earlier date, in which event it shall remain true and correct (as qualified, if applicable) as of such earlier date).
- (ii) <u>Covenants</u>. Buyer shall have performed and complied in all respects with all covenants and agreements required to be performed or complied with by it at or prior to the Closing Date.
- (iii) No Proceeding or Litigation. No litigation, action, suit, investigation, claim or proceeding challenging the legality of, or seeking to investigate, restrain, prohibit or materially modify, the transactions contemplated in this Agreement shall have been instituted and not settled or otherwise terminated, including without limitation under any foreign antitrust or competition Laws.
- (c) An incumbency certificate of each officer of Buyer who has executed or will execute a document in connection with the transactions contemplated herein on behalf of Buyer.
- (d) A copy of (i) the resolutions of the Board of Directors of Buyer authorizing and approving the transactions contemplated hereby and the execution and delivery of the Transaction Documents and (ii) Buyer's certificate of incorporation and bylaws, all certified by the corporate Secretary of Buyer to be true, correct, complete and in full force and effect and unmodified as of the Closing Date.
- (e) Evidence of the payment of the Base Cash Portion in the manner and the amount set forth in Section 1.2(a) and confirmation from UEI's transfer agent that the UEI Shares have been issued to Seller and recorded accordingly.
  - (f) Evidence that the UEI Shares represents the approximate percentage of the outstanding share capital of UEI set forth in Section 1.2.

#### **ARTICLE VI**

#### [INTENTIONALLY OMITTED]

#### ARTICLE VII

**SURVIVAL; INDEMNIFICATION** 

#### 7.1 Survival of Provisions; Assertion of Claims.

- (a) The representations and warranties of the Parties contained in this Agreement shall survive the Closing for a period of twelve (12) months after the Closing Date, save that (i) claims for breach of warranties in respect of Sections 2.1(a) (Organization and Standing; Power and Authority; Enforceability), 2.1(d) (Capitalization), 2.1(e) (Title to the Shares) and 2.1(v) (Taxes) shall be made within the period of forty two (42) months after the Closing Date and (ii) fraud and material misrepresentation shall be subject to statutory limitation period (the representations and warranties set forth in (i) and (ii) of this proviso the "Fundamental Representations"). The date upon which any representation, warranty contained herein shall terminate, if any, is referred to herein as the "Survival Date."
- (b) <u>Assertion of Claims</u>. Notwithstanding any provision to the contrary, no claim shall be brought under this ARTICLE VII unless the Indemnified Party, at any time prior to the applicable Survival Date, gives the Indemnifying Party(ies) a Claim Notice pursuant to Section 7.4 in respect of such claim or in respect of a Third Party Claim the existence of which might give rise to a claim.

#### 7.2 **Indemnity Obligations**

- (a) <u>Indemnity by Seller</u>. Subject to Section 7.3 hereof, Seller shall indemnify, defend and hold harmless Buyer and UEI (collectively, the "<u>Buyer</u> Indemnitees" provided that there shall not be any double claim against the Buyer) from and against any and all Damages arising out of or resulting from:
- (i) Any inaccuracy in or breach of any representation or warranty made by Seller in this Agreement (including the Disclosure Schedules) as of the date hereof and as of the Closing; and
- (ii) The failure of Seller to perform or observe any covenant, agreement or condition to be performed or observed by Seller, pursuant to this Agreement.
  - (b) Indemnity by Buyer. Buyer shall indemnify, defend and hold harmless Seller from and against any and all Damages arising out of or resulting from:
- (i) Any inaccuracy in or breach of any representation or warranty made by Buyer in this Agreement (including the Disclosure Schedules) as of the date hereof and as of the Closing; and

- (ii) The failure of Buyer to perform or observe any covenant, agreement or condition to be performed or observed by Buyer, pursuant to this Agreement.
- **7.3** <u>Limitations on Indemnification</u>. The Buyer Indemnitees shall not have the right to be indemnified pursuant to Section 7.2 unless and until the Buyer Indemnitees shall have incurred on a cumulative basis since the Closing Date aggregate Damages in an amount exceeding One Million Dollars (\$1,000,000) (the "<u>Basket</u>"), in which event the right to be indemnified shall be for Damages incurred by Buyer for the entire amount of such Damages, including the amount up to the Basket, up to Twelve Million and Five Hundred Dollars (\$12,500,000) (the "<u>Cap</u>"); <u>provided</u>, <u>however</u>, that in no event shall the limitations in respect of the Basket and the Cap set forth in this Section 7.3 apply to the rights of the Buyer Indemnitees to be indemnified for fraudulent or willful breaches or to the Buyer Indemnitees' rights to indemnification pursuant to Section 7.2 in respect of the Fundamental Representations or the indemnities set forth in Sections 7.2(a)(ii).
- **7.4** Notice of Claims. In the event an Indemnified Party wishes to make a claim for Damages (a "Claim") under Section 7.2, such Indemnified Party shall give a written notice (a "Claim Notice") to the Indemnifying Party. The Claim Notice shall state the nature and basis of such Claim and the amount thereof to the extent known. If such Claim relates to any assertion by any third party of any claim or of the commencement by any such third party of any action (a "Third Party Claim") with respect to which an Indemnifying Party(ies) is/are or may be obligated to provide indemnification, the Indemnified Party shall give a Claim Notice to the Indemnifying Party, within thirty (30) Business Days of receipt of notice of such Third Party Claim (or such shorter period as may be warranted under the circumstances, such as in the case of emergency proceedings or when a response to a notification must be given within a period in order to avoid a forfeiture of rights); provided, however, that no delay on the part of the Indemnified Party or Indemnified Parties shall relieve the Indemnifying Party or Indemnifying Parties (as the case may be) from any Liability hereunder so long as the Claim Notice is served prior to Survival Date.

#### 7.5 Third Party Claims.

(a) Initial Procedure. An Indemnifying Party shall have fifteen (15) Business Days after receipt of a Claim Notice in respect of a Third Party Claim to acknowledge in writing responsibility for the defense of such Third Party Claim and undertake, conduct and control, through counsel of its own choosing, which counsel shall be reasonably satisfactory to such Indemnified Party (or Indemnifying Parties, as the case may be), and at the Indemnifying Party's (or Indemnifying Parties, as the case may be) sole expense, the settlement or defense thereof; provided, however, that the Indemnifying Party, or Indemnifying Parties, as the case may be, shall not have the right to assume the defense of any Third Party Claim, notwithstanding the giving of such written acknowledgment, if (i) the Third Party Claim seeks only an injunction or other equitable relief, (ii) the Indemnified Party shall have been advised in writing by counsel that there are one or more legal or equitable defenses available to it which are different from or in addition to those available to the Indemnifying Party (or Indemnifying Parties, as the case may be), and, in the reasonable opinion of the Indemnified Party, counsel for the Indemnifying Party (or Indemnifying Parties, as the case may be) could not adequately represent the interests of the Indemnified Party because such interests could be in conflict with those of the Indemnifying

Party (or Indemnifying Parties, as the case may be) which written advice (or a summary thereof) shall be provided to the Indemnifying Party promptly after receipt by the Indemnified Party, (iii) in the reasonable determination of the Indemnified Party, which determination shall be provided in writing to the Indemnifying Party promptly after having made such action or proceeding involves, or could have a material effect on, any material matter beyond the scope of the indemnification obligation of the Indemnifying Party (or Indemnifying Parties, as the case may be) or (iv) the Indemnifying Party shall not have assumed the defense of the Third Party Claim in a timely fashion (which shall not in any event be less than 15 Business Days).

- (b) <u>Participation by Indemnified Party</u>. If the Indemnifying Party shall assume the defense of a Third Party Claim as provided herein, the Indemnifying Party shall permit the Indemnified Party to participate in the settlement or defense of such claim through counsel chosen by the Indemnified Party, provided that the fees and expenses of such counsel shall not be borne by the Indemnifying Party. No Indemnifying Party shall under any circumstances whatsoever consent to the entry of any judgment, pay or make any settlement of any claims without the prior written consent of the Indemnified Party (which consent shall not be unreasonably withheld or delayed).
- (c) <u>Assumption of Defense by Indemnified Party</u>. If the Indemnifying Party, or the Indemnifying Parties, as the case may be, does not or do not notify the Indemnified Party within the applicable period specified above after receipt of the Claim Notice that it or they (as the case may be) elect to undertake the defense of the Third Party Claim described therein, or the Indemnifying Party (or Indemnifying Parties, as the case may be) is or are otherwise restricted from so assuming the defense as a result of the application of the proviso to Section 7.5(a), (i) the Indemnified Party shall have the right to defend the Third Party Claim but shall not consent to the entry of any judgment, pay or settle such Third Party Claim without the prior written consent of the Indemnifying Party or the Indemnifying Parties, as the case may be, which consent shall not be unreasonably withheld, (ii) the Indemnifying Party (or Indemnifying Party Claim (including reasonable attorneys' fees and expenses), and (iii) the Indemnifying Party (or Indemnifying Parties, as the case may be) will remain responsible for any Damages the Indemnified Party may suffer resulting from, arising out of, relating to, in the nature of, or caused by the Third Party Claim to the extent provided in this ARTICLE VII.
- **7.6** Objection to Claim Notice. If any Indemnifying Party wishes to object to a Claim or Third Party Claim made in a Claim Notice, such Indemnifying Party shall give a written objection (an "Objection") to the Indemnified Party within thirty (30) Business Days after receipt of such Claim Notice expressing such Objection and explaining in detail and in good faith the basis thereof. The objecting Indemnifying Party and the Indemnified Party shall meet within fifteen (15) Business Days following receipt by the Indemnified Party of such Objection to agree on the rights of the respective parties with respect to each of such Claim or Third Party Claim. If such Parties should reach an agreement, a memorandum setting forth such agreement shall be prepared and signed by all such Parties. If such Parties do not reach such an agreement within such fifteen (15) Business Day period, then the matter shall be resolved in accordance with Section 8.1 hereof.

- **7.7** <u>Payment</u>. The payment of any sum due or payable by Seller under this Article VII shall (i) be made in immediately available funds by bank wire transfer to an account designated in writing by Buyer, the payment of which shall be secured by the Holdback Amount, in accordance with the terms and provisions of Section 1.2(a) (until the expiration of the Holdback Period in respect of claims where Claim Notices have been served prior to the Survival Date and subject to that portion of the Holdback Amount that has been segregated to cover any such pending Claims) and (ii) be paid as follows:
- (a) <u>Third Party Claims</u>. In connection with a Third Party Claim, no later than five (5) Business Days (a) after expiration of such fifteen (15) Business Day period set forth in Section 7.5(a) if an Indemnifying Party(ies) fail(s) to respond to the relevant Claim Notice, (b) after a final non appealable award against an Indemnifying Party(ies) has been rendered, (c) after final settlement of a Claim Notice has been reached, as determined by mutual agreement of the applicable Parties in accordance with Section 7.5(a) or (d) after an Indemnifying Party(ies) is/are determined to be under the obligation to pay the relevant Damages as resolved between Buyer and Seller(s) pursuant to Section 8.1.
- (b) <u>Direct Claims</u>. In connection with a direct (non Third Party) Claim, no later than five (5) Business Days (a) after expiration of such fifteen (15) Business Day period set forth in Section 7.5 (a) if and Indemnifying Party(ies) fail(s) to respond to the relevant Claim Notice, (b) after final settlement of a Claim Notice has been reached, as determined by mutual agreement of the applicable Parties in accordance with Section 7.6 or (c) after an Indemnifying Party(ies) is/are determined to be under the obligation to pay the relevant Damages as resolved between Buyer and Seller(s) pursuant to Section 8.1.
- (c) <u>Payment Recourse</u>. The Parties agree that the Buyer Indemnitees' rights to indemnification payments pursuant to Section 7.2(a) shall be made (i) until the release of the entire Holdback Amount in accordance with the terms of Section 1.2(a), from the Holdback Amount in the amount of such Damages incurred by Buyer; and (ii) following the release of the entire Holdback Amount, or if such Damages exceed in whole or in part the amount of cash available in the Holdback Amount by Seller to Buyer in the amount of such Damages incurred by Buyer which cannot be discharged through payment made by from the Holdback Amount.
- **7.8** No Limitation on Rights to Indemnification. Seller acknowledges that save as set out in the Transaction Documents (including the disclosure letter), no due diligence investigation, review and analysis performed by Buyer and its representatives, nor any knowledge of Buyer gained during such investigation, review and analysis that would provide Buyer with knowledge as to the accuracy or inaccuracy of the representations and warranties of Seller contained in ARTICLE II shall in any way limit or be deemed to limit the right of Buyer Indemnitees to indemnification hereunder.

#### ARTICLE VIII

#### **GENERAL**

- **8.1** Governing Law. This Agreement shall in all respects be governed by and construed in accordance with the laws of Hong Kong and the Parties hereto submit to the non-exclusive jurisdiction of the Hong Kong courts.
- **8.2** Schedules, Addenda and Exhibits. All schedules, addenda and exhibits attached to this Agreement, including without limitation the Disclosure Schedules, are incorporated herein and shall be part of this Agreement for all purposes.
  - 8.3 Amendments. This Agreement may be amended only by a writing executed by all of the Parties.
- **8.4** Entire Agreement. The Transaction Documents set forth the entire understanding of the Parties with respect to the subject matter hereof, and supersede all prior contracts, agreements, arrangements, communications, discussions, representations and warranties, whether oral or written, between the Parties, save in respect of the confidentiality undertaking by UEI to Seller dated September 13, 2010.
- **8.5** Assignment. This Agreement shall be binding upon and inure to the benefit of the successors and assigns of each Party, but, no rights, obligations or liabilities hereunder shall be assignable by Buyer without the prior written consent of Seller or by Seller without the prior written consent of Buyer, and any purported assignment in violation of this Section 8.5 shall be null and void *ab initio*.
- **8.6** Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, and all of which together will constitute one and the same instrument.
- **8.7** <u>Waivers</u>. Except as otherwise provided herein, either Party may waive in writing compliance by the other Parties hereto with any of the terms, covenants or conditions contained in this Agreement (except such as may be imposed by law). Any waiver by any Party of any violation of, breach of, or default under, any provision of this Agreement, by the other Party shall not be construed as, or constitute, a continuing waiver of such provision, or waiver of any other violation of, breach of or default under any other provision of this Agreement.
- **8.8** Third Parties. Nothing expressed or implied in this Agreement is intended, or shall be construed, to confer upon or give any Person other than Buyer and Seller any rights or remedies under or by reason of this Agreement.
- **8.9** <u>Headings</u>. The headings in this Agreement are solely for convenience of reference and shall not be given any effect in the construction or interpretation of this Agreement.
- **8.10** Gender and Number; Section and Article References. The masculine, feminine or neuter gender and the singular or plural number shall each be deemed to include the

others whenever the context so indicates. References to "\$" or "dollars" shall be to United States dollars. References to "HK\$" shall be to Hong Kong dollars. All references to Articles or Sections refer to Articles or Sections of this Agreement, unless otherwise stated.

- **8.11** <u>Time of Essence</u>. Time is of the essence for each and every provision of this Agreement, including without limitation Section 3.1.
- **8.12** <u>Interpretation.</u> No Party, nor its counsel, shall be deemed the drafter of this Agreement for purposes of construing the provisions of this Agreement, and all provisions of this Agreement shall be construed in accordance with their fair meaning, and not strictly for or against any Party.

**8.13** Notices. Any notice, request or other communication required or permitted hereunder shall be in writing and shall be deemed to have been duly given (a) when received if personally delivered, (b) within five (5) days after being sent by registered or certified mail, return receipt requested, postage prepaid, (c) within twelve (12) hours after being sent by telecopy, with confirmed answer back, or (d) within one (1) business day of being sent by priority delivery by established overnight courier, to the Parties at their respective addresses set forth below:

To Seller: CG International Holdings Limited

c/o Sogo Hong Kong Co. Ltd. 20/F, East Point Centre 555 Hennessy Road

Hong Kong Fax: 852 2831 4623

With a copy to: Richards Butler

20th Floor, Alexandra House

16-20 Chater Road Hong Kong Fax: 852 2810 0664 Attention: Ms. Janet Cheung

To Buyer: UEI Hong Kong Private Limited

Unit 2105-2107, Level 21, Tower 2, Grand Central Plaza 138 Shatin Rural Committee Road, Shatin, New Territories

Hong Kong

Attention: Richard A. Firehammer, Jr., Esq.

With a copy to: UEI Hong Kong Private Limited

c/o Universal Electronics 6101 Gateway Drive Cypress, California 90630

Attention: Paul D. Arling, Chairman

Fax: +1 (714) 820-1135

Universal Electronics Inc. 8350 Devon Court Chagrin Falls, Ohio 44023

Attention: Richard A. Firehammer, Jr., Esq.

Fax: +1 (440) 708-0721

Jones Day

555 South Flower Street, 50th Floor

Los Angeles, CA 90071

Attention: Chelsea A. Grayson, Esq.

Fax: +1 (213) 243-2539

To UEI: Universal Electronics Inc.

8350 Devon Court

Chagrin Falls, Ohio 44023

Attention: Richard A. Firehammer, Jr., Esq.

Fax: +1 (440) 708-0721

With a copy to: Jones Day

555 South Flower Street, 50th Floor

Los Angeles, CA 90071

Attention: Chelsea A. Grayson, Esq.

Fax: +1 (213) 243-2539

Any Party by written notice to the others given in accordance with this Section 8.13 may change the address or the Persons to whom notices or copies thereof shall be directed.

#### 8.14 Process Agents.

(a) Seller hereby irrevocably appoints Ms. Ivy Lam of 20/F, East Point Centre, 555 Hennessy Road, Hong Kong, as its agent to receive and acknowledge on its behalf service of any writ, summons, order, judgment or other notice of legal process in Hong Kong. If for any reason the agent named above (or its successor) no longer serves as agent of Seller in Hong Kong for this purpose, Seller shall promptly appoint a successor agent satisfactory to the other Parties, notify the other Parties thereof and deliver to the other Parties a copy of the new process agent's acceptance of appointment, provided that until the Parties receive the notification, it/they shall be entitled to treat the agent named above (or its said successor) as the agent of Seller for the purposes of this section. Seller agrees that any such legal process shall be sufficiently served on it if delivered to such agent for service at his address for the time being in Hong Kong whether or not such agent gives notice thereof to Seller.

(b) UEI hereby irrevocably appoints UEI Hong Kong Private Limited of the address set forth in Section 8.13 above as its agent to receive and acknowledge on its behalf service of any writ, summons, order, judgment or other notice of legal process in Hong Kong. If for any reason the agent named above (or its successor) no longer serves as agent of UEI in Hong Kong for this purpose, UEI shall promptly appoint a successor agent satisfactory to the other Parties, notify the other Parties thereof and deliver to the other Parties a copy of the new process

agent's acceptance of appointment, provided that until the Parties receive the notification, it/they shall be entitled to treat the agent named above (or its said successor) as the agent of UEI for the purposes of this section. UEI agrees that any such legal process shall be sufficiently served on it if delivered to such agent for service at his address for the time being in Hong Kong whether or not such agent gives notice thereof to UEI.

IN WITNESS WHEREOF, the Parties have caused their duly authorized representatives to execute this Agreement as of the date first written above.		
	SELLER:	
	CG INTERNATIONAL HOLDINGS LIMITED	
	By:	
	Name:	
	Title:	
	BUYER:	
	UEI HONG KONG PRIVATE LIMITED	
	By:	
	Name: Mark S. Kopaskie	
	Title: Director and Legal Representative	
	UNIVERSAL ELECTRONICS INC.	
	By:	
	By:Name: Paul D. Arling	
	Title: Chief Executive Officer	

[Signature Page to Stock Purchase Agreement]

#### ANNEX A

#### **Definitions**

- "Account" has the meaning set forth in Section 1.2(a).
- "Affiliate" means any Person that directly, or indirectly through one or more Persons, controls, is controlled by, or is under common control with, the Person specified. For purposes of this definition, "control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities, by contract or otherwise.
- "Agreement" has the meaning set forth in the Introduction.
- "Assets" means the tangible properties, the real properties and the business intellectual property, collectively.
- "Auditors" means Deloitte Touche Tohmatsu, Hong Kong.
- "Base Cash Portion" has the meaning set forth in Section 1.2.
- "Basket" has the meaning set forth in Section 7.3.
- "Books and Records" means all of the following that pertain to each of the Companies or their respective businesses and that are in the possession of any Company: books, records, manuals and other materials, accounting books and records, continuing property records for property, plant and equipment, files, computer tapes, disks, other storage media and records, advertising matter, catalogues, price lists, correspondence, mailing lists, lists of customers and suppliers, distribution lists, photographs, sales and promotional materials and records, purchasing materials and records, personnel records, employee handbooks, policies, rules or procedures, credit records, quality control records and procedures, research and development files, records, data, trademark files and disclosures, media materials and plates, sales order files and copies of litigation files and instruments relating to real property.
- "Business Day" means any day that is not a Saturday, Sunday or a day on which banking institutions in Los Angeles, California are not required to be open.
- "Business Intellectual Property" has the meaning set forth in Section 2.1(p).
- "Buyer" has the meaning set forth in the Introduction.
- "Buyer Indemnitees" has the meaning set forth in Section 7.2(a).
- "BVI Companies" means CG Group and CG Asia.
- "Cap" has the meaning set forth in Section 7.3.
- "CG Asia" has the meaning set forth in Section 2.1(a)(ii).

- "CG Asia Shares" has the meaning set forth in Section 2.1(d)(ix).
- "C.G. Development" means C.G. Development Limited, a limited liability company organized under the Laws of Hong Kong.
- "C.G. Development Shares" has the meaning set forth in Section 2.1(d)(iii).
- "CG Group" means CG Group Limited, a company organized under the Laws of the British Virgin Islands.
- "CG Group Shares" has the meaning set forth in Section 2.1(d)(vii).
- "C.G. Technology" means C.G. Technology Limited, a limited liability company organized under the Laws of Hong Kong.
- "C.G. Technology Shares" has the meaning set forth in Section 2.1(d)(ii).
- "C.G. Timepiece" has the meaning set forth in Section 2.1(a)(ii).
- "C.G. Timepiece Shares" has the meaning set forth in Section 2.1(d)(viii).
- "Claim" has the meaning set forth in Section 7.4.
- "Claim Notice" has the meaning set forth in Section 7.4.
- "Closing" has the meaning set forth in Section 1.1(b).
- "Closing Date" has the meaning set forth in Section 1.1(b).
- "Companies" means Enson and the Subsidiaries and "Company" shall be construed accordingly.
- "Company Earnings" means Enson's consolidated profit before tax for the fiscal year ending March 31, 2011 excluding the following items: investment income, other income, other expenses, other gains and losses and interest expenses (which items shall have the same definition as per Enson's audited consolidated financial statements for the fiscal year ended March 31, 2010) in accordance with the Company Earnings Principles and adjusted in accordance with the agreed adjustment as set out in Annex C, excluding net sales to UEI multiplied by standard gross margin of UEI's products equal to 16% less allocated operating expenses. Allocated operating expenses will be determined by the following equation: consolidated operating expenses as a percentage of consolidated net sales multiplied by 75% multiplied by net sales to UEI.
- "Company Earnings Principles" means those accounting principles, practices, methodologies and policies as set out in the latest audited consolidated financial statements of Enson.
- "Commission" means the United States Securities and Exchange Commission.
- "Consents" means consents, novations, approvals, authorizations, requirements (including filing and registration requirements), waivers and agreements.

- "Contested Claim" has the meaning set forth in Section 1.2(a)(i).
- "<u>Damages</u>" means any and all claims, actions, losses, obligations, costs, expenses, settlement payments, awards, damages, judgments, fines, penalties and other Liabilities of any kind or nature whatsoever, including without limitation reasonable attorneys', accountants' and experts' fees.
- "Direct Subsidiary Shares" has the meaning set forth in Section 2.1(e)(ii).
- "Disclosure Schedules" has the meaning set forth in Section 2.1.
- "DOJ" means the United States Department of Justice.
- "Employee Plan" or "Employee Plans" has the meaning set forth in Section 2.1(s).
- "Enson" has the meaning set forth in the Introduction.
- "Environmental Laws" means the applicable Law of any Governmental Authority relating to the prevention of pollution, regulating transportation, storage, discharge or emission of Hazardous Materials, remediation of contamination, protection of natural resources or the environment, preservation of environmental quality or the protection of human health or worker safety.
- "Environmental Matters" means matters relating to the environment, the protection of human health and worker safety, or the generation, use, storage, handling or disposal of Hazardous Materials.
- "Expenses" has the meaning set forth in Section 4.1.
- "Financial Statements" has the meaning set forth in Section 2.1(a)(i).
- "First Determination Date" means the earlier of (i) in the event that neither Seller nor Enson provides a Dispute Notice pursuant to Section 1.3(a)(i), five (5) Business Days after the date the Auditor delivers its report to Enson and Seller, (ii) the date Enson and Seller otherwise agree in writing upon the Auditor's report or (iii) the date a final, nonappealable judgment has been rendered pursuant to Section 8.1 resolving all disputed issues in respect of the report as raised in the Dispute Notice.
- "First Reference Amount" has the meaning set forth in Section 1.3(d).
- "FTC" means the Federal Trade Commission.
- "Fundamental Representation" has the meaning set forth in Section 7.1(a).
- "Gemstar (China)" has the meaning set forth in Section 2.1(a)(iii).
- "Gemstar (China) Interests" has the meaning set forth in Section 2.1(d)(iv).
- "Gemstar Companies" has the meaning set forth in Section 2.1(a)(iii).

- "Gemstar Polyfirst" means Gemstar Polyfirst Limited, a limited liability company organized under the Laws of Hong Kong.
- "Gemstar Polyfirst Shares" has the meaning set forth in Section 2.1(d)(vi).
- "Gemstar (Yangzhou)" has the meaning set forth in Section 2.1(a)(iii).
- "Gemstar (Yangzhou) Interests" has the meaning set forth in Section 2.1(d)(iv)2.1(d)(v).
- "Governmental Authority" means any government or state or any subunit thereof, multinational, foreign or domestic, federal, state or local governmental entity or municipality or subdivision thereof or any authority, department, commission, board, bureau, agency, court of any government or state or similar body or instrumentality, any governmental tribunal or other instrumentality.
- "<u>Hazardous Materials</u>" means each substances designated as a hazardous waste, hazardous substance, hazardous material, pollutant, contaminant or toxic substance under any Environmental Law including, without limitation, any asbestos, mold, lead-based paint, polychlorinated biphenyls, urea formaldehyde foam insulation, and petroleum or any fraction of petroleum.
- "Holdback Amount" has the meaning set forth in Section 1.2(a).
- "Holdback Period" has the meaning set forth in Section 1.2(a).
- "Hong Kong" means the Hong Kong Special Administrative Region of the People's Republic of China.
- "HK\$" means Hong Kong dollars, the lawful currency of Hong Kong.
- "HKFRS" means the Hong Kong Financial Reporting Standards.
- "Hong Kong Companies" means C.G. Technology, C.G. Development, Gemstar Polyfirst and C.G. Timepiece.
- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.
- "Indebtedness" means all (i) liabilities and obligations for borrowed money or evidence by notes, bonds (including surety bonds and similar obligations) or similar instruments, (ii) obligations in respect of deferred purchase price of property or services (other than current trade liabilities incurred in the ordinary course of business consistent with past practice), (iii) obligations in respect of capitalized leases, (iv) obligations under interest rate cap agreements, interest rate swap agreements, foreign currency exchange contracts or other hedging contracts and (v) any guarantee of the obligations of another Person with respect to any of the foregoing.
- "Indemnified Party" means any Party entitled to be indemnified pursuant to ARTICLE VII hereof.

"Indemnifying Party" means any Party obligated to provide indemnification pursuant to ARTICLE VII hereof.

"Insolvent" means: (i) any event in which either Seller is unable to pay its debts as they mature; (ii) the sum of the assets and properties of a Seller at a fair valuation are less than its debts and other Liabilities; (iii) any event in which a Seller is required to make an assignment for the benefit of creditors; (iv) any event in which a Seller shall be required to file a voluntary petition in bankruptcy, or shall be adjudicated a bankrupt or insolvent, or shall be required to file any petition or answer seeking for itself any reorganization, arrangement, composition, readjustment, liquidation, dissolution or similar relief under any present or future statute, law or regulation, or shall be required to file any answer admitting or not contesting the material allegations of a petition filed against it in any such proceeding; or (v) any event in which a Seller shall be required to seek or consent to or acquiesce in the appointment of any trustee, receiver or liquidator of it or of all or any substantial part of its properties.

"Intellectual Property" means all intellectual property rights and related priority rights, whether protected, created or arising under the laws of the United States or any other jurisdiction or under any international convention, including all (i) patents and patent applications, including all continuations, divisionals, continuations-in-part, and provisionals and patents issuing on any of the foregoing, and all reissues, reexaminations, substitutions, renewals and extensions of any of the foregoing (collectively, "Patents"), (ii) trademarks, service marks, trade names, trade dress, logos, corporate names and other source or business identifiers, together with the goodwill associated with any of the foregoing, and all applications, registrations, renewals and extensions of any of the foregoing (collectively, "Marks"), (iii) Internet domain names, (iv) copyrights, works of authorship, including rights in databases, data collections and moral rights, and all registrations, applications, renewals, extensions and reversions of any of the foregoing (collectively, "Copyrights"), (v) mask works and mask sets, and all applications and registrations of any of the foregoing (collectively, "Mask Works") and (vi) confidential and proprietary information, trade secrets and non-public discoveries, concepts, ideas, research and development, technology, know-how, formulae, inventions, compositions, processes, techniques, technical data and information, procedures, semiconductor device structures (including gate structures, transistor structures, memory cells or circuitry, vias and interconnects, isolation structures and protection devices), circuit block libraries, designs (including circuit designs and layouts), drawings, specifications, databases, data collections and other information, including customer lists, supplier lists, pricing and cost information, and business and marketing plans and proposals, in each case excluding any rights in respect of any of the foregoing that comprise or are protected by Patents (collectiv

"Knowledge" means, with respect to Seller, the knowledge of the directors, officers and executives of Seller after due inquiry, which knowledge will include the knowledge such Persons would have had after due inquiry of the responsible officers and employees of Seller and the Companies and their counsel and accountants.

"Law" means any applicable law, order, statute, rule, regulation, ordinance or other act of a legally binding nature of domestic and foreign federal, state and local Governmental Authorities in force and effect from time to time.

- "Lease" has the meaning set forth in 2.1(g)(i).
- "Leased Real Property" has the meaning set forth in 2.1(g)(i).
- "<u>Liability</u>" means any liability or obligation, whether known or unknown, asserted or unasserted, absolute, fixed or contingent, accrued or unaccrued, liquidated or unliquidated and whether due or to become due, regardless of when asserted, including without limitation arising out of litigation, actions, suits, claims or proceedings.
- "Lien" means any and all liens, equities, claims, prior assignments, mortgages, charges, security interests, pledges, options, conditional sales contracts, collateral security arrangements and other title retention arrangements, whether voluntarily incurred or arising by operation of law.
- "Material Customer" being the ten (10) largest customers of the Companies during the six (6) month period ended September 30, 2010.
- "Material Supplier" being the ten (10) largest suppliers of the Companies during the six (6) month period ended September 30, 2010.
- "<u>Net Assets</u>" means the amount by which, as of the Closing Date (i) consolidated assets of Enson exceed (ii) consolidated liabilities of Enson, in each case determined in accordance with the Net Assets Principles and adjusted in accordance with the agreed adjustment as set out in <u>Annex C</u>.
- "Net Assets Principles" means those accounting principles, practices, methodologies and policies as set out in the latest audited consolidated financial statements of Enson.
- "Non Competition Agreement" has the meaning set forth in Section 5.1(g).
- "Objection" has the meaning set forth in Section 7.6.
- "Owned Real Property" has the meaning set forth in 2.1(g)(i).
- "Party" or "Parties" has the meaning set forth in the Introduction.
- "Permit" means any license, permit, approval, authorization, variance, waiver or consent issued to any Company by any Governmental Authority.
- "<u>Permitted Liens</u>" means (i) Liens for Taxes not delinquent or the validity of which is being contested in good faith by appropriate proceedings and as to which adequate reserves have been established on the face of the Financial Statements reflecting the full amount of such contested Taxes, (ii) statutory landlord's, mechanic's, materialmen's, carrier's, workmen's, repairmen's or other similar Liens arising or incurred in the ordinary course of business and for amounts which are not delinquent and which are set forth on the face of the Financial Statements and (iii) with respect to any Real Property, easements, covenants and other restrictions of record which do not materially affect the use or value of such Real Property.

- "<u>Person</u>" means and shall be construed broadly and includes without limitation a or an individual, partnership, corporation, limited liability company, association, joint stock company, trust, joint venture, unincorporated organization or Governmental Authority.
- "PRC" means the People's Republic of China, for the purpose of this Agreement, does not include Hong Kong, Macau and Taiwan.
- "Prime Rate" means a fixed annual rate equal to the "Prime Rate" as reported in The Wall Street Journal.
- "Purchase Price" has the meaning set forth in Section 1.2.
- "Real Property" has the meaning set forth in 2.1(g)(i).
- "<u>Related Person</u>" means a member of the family of a Person or a stockholder, director or executive of a Person, or any entity (including trusts or family limited liability companies, etc.) owned by or for the benefit of a Person or a stockholder, director or executive of a Person or members of the family thereof. For this purpose, the term family means spouses, parents, siblings, children or trusts for the benefit of any such Person.
- "Release" means any spilling, leaking, pumping, pouring, emitting, discharging, injecting, escaping, leaching, dumping or disposing of a Hazardous Material into the environment.
- "Release Date" has the meaning set forth in Section 1.2(a).
- "Second Determination Date" means the earlier of (i) in the event that neither Seller nor Enson provides a Dispute Notice pursuant to Section 1.4(a), five (5) Business Days after the date the Auditor delivers the Audited Accounts and the Companies Earnings to Enson and Seller, (ii) the date Enson and Seller otherwise agree in writing upon the Audited Accounts and the Company Earnings or (iii) the date a final, nonappealable judgment has been rendered pursuant to Section 8.1 resolving all disputed issues in respect of the Audited Accounts and the Company Earnings as raised in the Dispute Notice.
- "Second Reference Amount" has the meaning set forth in Section 1.4(d).
- "Segregated Funds" has the meaning set forth in Section 1.2(a)(i).
- "Seller" has the meaning set forth in the Introduction.
- "Seller's Account" has the meaning set forth in Section 1.2(a).
- "Shares" means that number of shares of Enson that constitute all of the issued share capital of Enson as at the Closing Date.
- "Stock Portion" has the meaning set forth in Section 1.2.

- "Subsidiary" or "Subsidiaries" means the following entities: Gemstar Polyfirst, C.G. Technology, C.G. Development, Gemstar (China), Gemstar (Yangzhou), CG Group, C.G. Timepiece and CG Asia.
- "Survival Date" has the meaning set forth in Section 7.1(a).
- "Tangible Property" has the meaning set forth in Section 2.1(f).
- "Tax" or "Taxes" means all net income, gross income, gross receipts, sales, use, personal property, real property, ad valorem, transfer, franchise, profits, license, withholding, payroll, employment, excise, severance, stamp, occupation, premium, property or windfall profits taxes, customs duties, together with any interest and any penalties, additions to tax or additional amounts imposed by any taxing authority (domestic or foreign).
- "<u>Tax Return</u>" means any declaration, statement, report, return or other document or information required to be filed or supplied with respect to Taxes to any Governmental Authority including any amendment thereof.
- "Third Party Claim" has the meaning set forth in Section 7.4.
- "Transaction Documents" means this Agreement and all other agreements and instruments executed or delivered by Seller in connection herewith.
- "Transfer Taxes" has the meaning set forth in Section 3.6.
- "UEI" has the meaning set forth in the Introduction.
- "UEI Shares" has the meaning set forth in Section 1.2.

# ANNEX B [INTENTIONALLY OMITTED]

#### ANNEX C

#### <u>Agreed Adjustments to Auditors on the Calculation</u> <u>of the Net Assets and Earnings Adjustment</u>

#### Property, plant and equipment

No item will be derecognised through Mar 31, 2011 including 7 ultra sonic welding machines, 4 the temporary warehouses in Panyu and the 2 dormitory buildings in Yangzhou.

#### Leasehold land and buildings under development for future owner-occupied purpose

Depreciation on the new staff quarter and warehouse in Gemstar Yangzhou will start when the completion permit is issued by relevant authority.

#### Revenue recognition

Sale of materials and spare parts and samples is recognised when the invoice is despatched to customers.

Tooling income is recognised when services are provided and accepted by customers. This arises when payment is received from customers.

Interest income from a financial asset is accrued on a time basis, by reference to the principal outstanding and at the effective interest rate offered by relevant bank, which is the rate that exactly discounts the estimated future cash receipts through the expected life of the financial asset to that asset's net carrying amount on initial recognition.

#### Club debenture

No impairment loss as at March 31, 2011.

### Impairment loss on tangible and intangible assets

No impairment loss through March 31, 2011.

#### Financial instruments

Financial assets and financial liabilities are stated at cost except for accounts receivables.

#### Allowance for trade receivables

Allowance for trade receivables are provided on the following basis:-

a. No adjustment on tooling income.

b. trade receivables (UEI excluded) — 20% of due over 120 days

#### Inventory valuation method and impairment

The followings are the basis for identifying slow-moving and obsolete inventories:-

- (1) Raw materials and work in progress 20% materials held between 7 to 12 months 50% for over 12 months
- (2) Finished goods 50% of finished goods held over 12 months

#### Provision for Social insurance, Housing Fund and Severance Payment

The provision and payment basis of social insurance and housing fund are same as those at August 31, 2010.

#### Estimated bonus (other than double pay)

Not more than HK\$4 million for the year ended Mar 31, 2011.

#### Others

Any administrative or overhead expenses allocated from the head office or other units of UEI and other expenses which will not be incurred by the Companies but the consolidation of the Companies into UEI shall be disregarded.

Any claims (including but not limited to claims from workers and employees and claims from relevant authority) in respect of PRC labour issues shall not be deductible from the current year operating income so long as the same could be covered by the pool of provisions made by the Company in the prior years.

## EXHIBIT A

## [Intentionally omitted.]

## EXHIBIT B

## Form of Non Competition Agreement

— Attached —

## EXHIBIT C

## Form of Key Employee Employment Agreement

— Attached —

# EXHIBIT D Form of Certificate by Officer



Contacts: Paul Arling (UEI) 714.820.1000 Kirsten Chapman (IR Agency) 415.433.3777

#### Universal Electronics Acquires Leading Remote Control Company

- - Increases presence in rapidly growing global markets and broadens UEI's customer list - - - Management to hold conference call at 4:30 pm ET today -

CYPRESS, CA — November 4, 2010 — Universal Electronics Inc. (UEI), (NASDAQ: UEIC) has acquired China based Enson Assets Limited from CG International Holdings Limited for a net purchase value of approximately \$110 million. Enson, through its wholly owned subsidiaries located in China, C.G. Development Limited and C.G. Technology Limited is a leading designer, marketer and manufacturer of remote controls for original equipment manufacturers (OEMs)

"This acquisition further strengthens UEI's leadership position in wireless control technology as it positions us to benefit from very promising international growth opportunities and significantly increases our market share in OEMs," stated Paul Arling, UEI's Chairman and CEO. "We have a six-year history working with the C.G. companies, and we are excited to welcome their employee base, including experienced management and engineering groups, to UEI. This transaction is also expected to increase our platform for growth in Asia, one of the world's most rapidly growing regions where we expect to expand our market share in OEMs and subscription broadcasting. This acquisition will also strengthen our customer list with key industry-leading consumer electronics companies including Sony, Panasonic and Toshiba."

"In addition to the strategic value gained by the transaction, the financial terms of the acquisition as defined in the definitive agreement are extremely attractive for UEI, and we expect the acquisition to be immediately accretive," stated Bryan Hackworth, UEI's CFO. "While we have historically given guidance for the coming fiscal year in February, our current forecast shows that this acquisition will add at least \$140 million to our sales total and at least \$20 million in operating income, leading to more than \$1.00 in EPS accretion in fiscal year 2011. We will, as has been our custom, provide greater detail to our overall forecast when we disclose our fourth quarter 2010 results in February 2011."

The transaction is valued at approximately \$110 million. To purchase C.G., excluding transaction and related expenses, UEI will deliver 1.46 million shares of common stock and pay \$95 million in cash. UEI will receive net assets of \$68.5 million, including approximately \$15 million in cash at closing. Post transaction, UEI expects to have a cash and debt balance of \$35 million each. The financing used for the transaction will carry a variable interest rate, which is currently at approximately 1.8%.

#### **Third Quarter 2010 Conference Call Information**

Management will host a conference call at 4:30 p.m. ET, Thursday, November 4, 2010 to discuss and answer questions about its third quarter 2010 earnings results as well as today's announcement of the acquisition of Enson Assets Limited.

To access the call in the U.S. please dial 877-655-6895 and for international calls dial 706-758-0299 approximately 10 minutes prior to the start of the conference. The conference ID is 18361101. The conference call will also be broadcast live over the Internet and available for replay for one year at <a href="https://www.uei.com">www.uei.com</a>. In addition, a replay of the call will be available via telephone for two business days, beginning two hours after the call. To listen to the replay, in the U.S., please dial 800-642-1687 and internationally, 706-645-9291. Enter access code 18361101.

#### **About Enson Assets Limited**

Enson Assets Limited, through its wholly-owned subsidiaries C.G. Development Limited and C.G. Technology Limited, is principally engaged in the design, development, manufacture and sale of remote control units. The manufacturing business commenced with the establishment of the Panyu Plant in 1992. C.G. is a global market leader in the manufacturing of remote control units with a renowned global customer base, which is comprised of industry leading brands of home appliance and consumer electronic companies.

#### **About Universal Electronics Inc.**

Founded in 1986, Universal Electronics Inc. (UEI) is the global leader in wireless control technology for the connected home. UEI designs, develops, and delivers innovative solutions that enable consumers to control entertainment devices, digital media, and home systems. The company's broad portfolio of patented technologies and database of infrared control software have been adopted by many Fortune 500 companies in the consumer electronics, subscription broadcast, and computing industries. UEI sells and licenses wireless control products through distributors and retailers under the One For All® brand name. UEI also delivers complete home control solutions in the professional custom installation market under the brand name Nevo®. For additional information, please visit our website at <a href="https://www.uei.com">www.uei.com</a>.

#### **Safe Harbor Statement**

This press release contains forward-looking statements that are made pursuant to the Safe-Harbor provisions of the Private Securities Litigation Reform Act of 1995. Words and expressions reflecting something other than historical fact are intended to identify forward-looking statements. These forward-looking statements involve a number of risks and uncertainties, including the ability of the Company to successfully integrate the operations of Enson and its subsidiaries into our operations, the failure of Enson to perform in accordance with our expectations; our inability to obtain orders or maintain our order volume with new and existing customers; general economic conditions; and other factors described in the Company's filings with the U.S. Securities and Exchange Commission. The actual results that the Company achieves may differ materially from any forward looking statement due to such risks and uncertainties. The Company undertakes no obligations to revise or update any forward-looking statements in order to reflect events or circumstances that may arise after the date of this release.



Contacts: Paul Arling (UEI) 714.820.1000 Kirsten Chapman (IR Agency) 415.433.3777

#### Universal Electronics Reports Third Quarter 2010 Financial Results

- - Also Announces Acquisition of Leading Remote Control Company Under Separate Release Issued Today -

CYPRESS, CA — November 4, 2010 — Universal Electronics Inc. (UEI), (NASDAQ: UEIC) reported financial results for the third quarter ended September 30, 2010.

"We are pleased to report our third quarter results were generally in line with our expectations," stated Paul Arling, UEI's Chairman and CEO. "Our business faced some headwinds in the developed markets, most specifically in North American consumer electronics, but fared particularly well in developing economies. Looking forward, we couldn't be more excited about the future of our company, due in large part to the acquisition of Enson Assets Limited, including its subsidiaries C.G. Development Limited and C.G. Technology Limited, which we also announced this afternoon."

Arling added, "We expect this acquisition to place us squarely in the lead as the provider of wireless control devices into the consumer electronics market. Additionally, this acquisition significantly expands our product range, which will enable us to further penetrate the growing Asian and Latin American subscription broadcast markets. Further, we have over six years of experience working with the C.G. companies and know them well. They have consistently been a top supplier to UEI, due to its top-notch management team and experienced engineering group, whom we welcome to the UEI team."

#### Financial Results for the Quarter Ended September 30: 2010 Compared to 2009

- Net sales were \$79.0 million, compared to \$83.2 million.
  - § Business Category revenue was \$66.2 million, compared to \$67.0 million. The Business Category contributed 84% of total net sales, compared to 81%.
  - § Consumer Category revenue was \$12.8 million, compared to \$16.2 million. The Consumer Category contributed 16% of total net sales, compared to 19%.
- Gross margins were 32.6%, compared to 31.3%.
- Total operating expenses were \$19.2 million, compared to \$19.4 million.
- Operating income was \$6.6 million, compared to \$6.6 million.
- Interest expense was \$1,000, compared to interest income of \$110,000.
- Net income was \$4.7 million, or \$0.34 per diluted share, compared to \$4.2 million, or \$0.30 per diluted share.
- At September 30, 2010, cash and cash equivalents was \$73.0 million.

Bryan Hackworth, UEI's CFO, said: "While third quarter 2010 sales were down slightly relative to the third quarter of 2009, we improved gross margins to 32.6% of sales and lowered our operating expenses. As a result, we delivered operating margin improvement, increasing from 8.0% in the third quarter of 2009 to 8.3% in the third quarter of 2010."

#### Nine-month Period Ended September 30: 2010 Compared to 2009

- Net sales were \$229.3 million, compared to \$232.6 million.
- Gross margins were 32.8%, compared to 31.4%.
- Total operating expenses were \$58.6 million, compared to \$59.1 million.
- Net income was \$11.3 million, or \$0.81 per diluted share, compared to \$8.8 million, or \$0.63 per diluted share.

#### **Financial Outlook**

Today, it is expected the C.G. companies will add at least \$140 million in annual revenue, primarily in the OEM channel, to UEI's business in 2011, and at least \$20 million in operating income. The acquisition is expected to decrease the consolidated gross margin percentage by approximately 2 percentage points; however, the consolidated operating margin percentage is expected to increase by 2 percentage points. Overall, the acquisition is expected to result in at least \$1.00 in EPS accretion in 2011. The company expects to provide greater detail to its overall forecast when it discloses its fourth quarter 2010 results in February 2011.

Including the C.G. acquisition in the forecast and based on historical results, fourth quarter of 2010 net sales are expected to range between \$102.0 million and \$108.0 million, compared to \$84.9 million contributed from UEI in the fourth quarter of 2009 and approximately \$13.9 million in sales from C.G. for the comparable period corresponding to the acquisition. The comparable period for C.G. is defined as sales from November 4th to December 31st, net of sales to UEI.

The company anticipates gross margins for the fourth quarter of 2010 to be approximately 31.0% of sales, plus or minus one point. For the fourth quarter of 2010, operating expenses are expected to range from \$23.6 million to \$24.4 million, including approximately \$1 million expenses related to the acquisition. Earnings per diluted share for the fourth quarter of 2010 are expected to range from \$0.43 to \$0.49. Earnings per diluted share were \$0.42 in the fourth quarter of 2009.

Full year 2010 net sales including the C.G. acquisition are expected to range between \$331.3 million and \$337.3 million, compared to \$317.6 million from UEI in 2009 and approximately \$13.9 million from C.G. for the aforementioned comparable period. Earnings per diluted share for 2010 are expected to range from \$1.24 to \$1.30. Earnings per diluted share were \$1.05 in 2009.

#### **Conference Call Information**

UEI's management team will hold a conference call today, Thursday, November 4, 2010 at 4:30 p.m. ET / 1:30 p.m. PT, to discuss its third quarter 2010 earnings results, review the quarterly activity and answer questions. To access the call in the U.S. please dial 877-655-6895 and for international calls dial 706-758-0299 approximately 10 minutes prior to the start of the conference. The conference ID is 18361101. The conference call will also be broadcast live

over the Internet and available for replay for one year at <a href="www.uei.com">www.uei.com</a>. In addition, a replay of the call will be available via telephone for two business days, beginning two hours after the call. To listen to the replay, in the U.S., please dial 800-642-1687 and internationally, 706-645-9291. Enter access code 18361101.

#### **About Universal Electronics Inc.**

Founded in 1986, Universal Electronics Inc. (UEI) is the global leader in wireless control technology for the connected home. UEI designs, develops, and delivers innovative solutions that enable consumers to control entertainment devices, digital media, and home systems. The company's broad portfolio of patented technologies and database of infrared control software have been adopted by many Fortune 500 companies in the consumer electronics, subscription broadcast, and computing industries. UEI sells and licenses wireless control products through distributors and retailers under the One For All® brand name. UEI also delivers complete home control solutions in the professional custom installation market under the brand name Nevo®. For additional information, please visit our website at <a href="https://www.uei.com">www.uei.com</a>.

#### **About Enson Assets Limited**

Enson Assets Limited, through its wholly-owned subsidiaries C.G. Development Limited and C.G. Technology Limited is principally engaged in the design, development, manufacture and sale of remote control units. The manufacturing business commenced with the establishment of the Panyu Plant in 1992. C.G. is a global market leader in the manufacturing of remote control units with a renowned global customer base, which is comprised of industry leading brands of home appliance and consumer electronics companies.

#### **Safe Harbor Statement**

This press release contains forward-looking statements that are made pursuant to the Safe-Harbor provisions of the Private Securities Litigation Reform Act of 1995. Words and expressions reflecting something other than historical fact are intended to identify forward-looking statements. These forward-looking statements involve a number of risks and uncertainties, including the ability of the Company to successfully integrate the operations of Enson and its subsidiaries into our operations, the failure of Enson to perform in accordance with our expectations; the continued development of innovative control solutions and technologies that are accepted by our customers and consumers; the continued leveraging of the Company's fixed cost structure resulting in increased profitability and cash flow; general economic conditions; and other factors described in the Company's filings with the U.S. Securities and Exchange Commission. The actual results that the Company achieves may differ materially from any forward looking statement due to such risks and uncertainties. The Company undertakes no obligations to revise or update any forward-looking statements in order to reflect events or circumstances that may arise after the date of this release.

— Tables Follow —

## UNIVERSAL ELECTRONICS INC. CONSOLIDATED BALANCE SHEETS

(In thousands, except share-related data) (Unaudited)

Carrent assets:		September 30, 2010	December 31, 2009
Cash and cash equivalents         \$ 23,447         \$ 29,166           Term deposit         49,536         49,266           Accounts receivable, net         57,990         64,392           Inventories, net         44,615         49,347           Prepaid expenses and other current assets         1,934         2,428           Income tax receivable         480         —           Deferred income taxes         2,938         3,016           Total current assets         180,600         189,040           Equipment, furniture and fixtures, net         10,913         199,000           Goodwill         11,369         13,724           Intangible assets, net         11,323         11,527           Other assets         7,573         7,837           Total assets         7,573         7,837           Total come taxes         7,573         7,837           Total come taxes         6,592         6,028           Accured income taxes         6,692         6,028           Accured comps, rebates and royalties         5,550         4,619           Accured compensation         5,650         4,619           Other accured expenses         9         6,928           Total current liabilities	ASSETS		
Term deposit         49,356         49,246           Accounts receivable, net         57,999         64,392           Inventories, net         44,615         40,947           Prepaid expenses and other current assets         1,594         2,423           Income tax receivable         480         —           Deferred income taxes         180,000         189,040           Equipment, furniture and fixtures, net         10,913         9,990           Coodwill         13,609         13,724           (Income taxes)         17,223         11,522           Other assets         757         1,148           Other assets         7,833         7,837           Total assets         7,833         7,837           Total assets         7,833         7,837           Account spayable         \$30,909         \$39,514           Accrued income taxes         6,692         6,028           Accrued income taxes         6,698         8,539           Total current isabilities         9         6,098           Lise and income taxes         1,594         6,198           Other accrued expenses         6,089         8,595           Accrued income taxes         1,550         4,519			
Accounts receivable, net			
Hometories, net   44,615   40,947   Prepaid expenses and other current assets   1.594   2,422   Income tax receivable   480   -426   Perferred income taxes   180,600   189,040   Equipment, furniture and fixtures, net   10,913   9,990   Goodwill   13,609   13,724   Intangible assets, net   11,323   11,572   Intangible assets, net   11,323   7,775   1,144   Deferred income taxes   7,853   7,833   Total assets   7,853   7,835   Total assets   7,853   7,835   Total assets   7,853   7,835   Total assets   8,800   8,800    ELIABILITIES AND STOCKHOLDERS' EQUITY    Current liabilities   5,809   8,39,514   Accounts payable   8,30,692   6,028   Accrued compensation   6,692   6,028   Accrued compensation   6,692   6,028   Accrued compensation   6,565   6,195   Other ascrued expenses   6,688   6,539   Total current liabilities   40,339   6,1954   Income tax payable   1,348   1,348   Other long-term liabilities   1,348   1,348   Other long-term liabilities   7,8	Term deposit		
Prepaid expenses and other current assets         1,994         2,423           Income tax receivable         2,938         3,016           Total current assets         180,0600         189,040           Godwill         13,609         13,724           Godwill         13,609         13,724           Intangible assets, net         113,609         13,724           Other assets         7,57         1,144           Deferred income taxes         7,53         7,837           Total assets         7,53         7,837           Accounts payable         \$3,969         \$3,514           Accrued asset discounts, rebates and royalties         6,692         6,028           Accrued compensation         6,692         6,028           Accrued compensation         5,693         8,339           Total current liabilities         49,39         61,594           Long-term liabilities         1,59         1,594           Long-term liabilities         1,59         1,59           Long-term liabilities         1,59         1,51           Deferred income taxes         1,59         1,59           Deferred income taxes         1,59         1,59           Total liabilities         2,59			
Income tax receivable		44,615	40,947
Deferred income taxes         2,938         3,016           Total current assets         100,600         180,000           Equipment, firmiture and fixtures, net         10,913         9,909           Goodwill         11,302         11,722           Intangible assets, net         15,50         11,424           Other assets         7,853         7,833           Total assets         7,853         7,837           **Contract assets         225,055         \$23,307           **Contract assets         30,909         \$39,514           **Accounts payable         6,692         6,028           **Accured compensation         6,692         6,028           **Accured compensation         6,692         6,028           **Other accured expenses         6,082         6,503           **Total current liabilities         1,348         8,339           **Total current liabilities         1,348         1,348           **Deferred income taxes         1,348         1,348           Other long-term liabilities         7,89         1,22           Total liabilities         1,348         1,348           Other long-term liabilities	Prepaid expenses and other current assets	1,594	2,423
Total current assets         180,600         189,040           Equipment, furnitures, net         10,913         3,990           Goodwill         13,609         13,722           Intangible assets, net         11,323         11,572           Other assets         7,853         7,833           Total assets         7,853         7,833           Total assets         8225,055         \$233,307           LIABILITIES AND STOCKHOLDERS' EQUITY           Current liabilities:           Accrued siales discounts, rebates and royalties         6,992         6,028           Accrued carrent cincome taxes         6,692         6,028           Accrued compensation         5,650         4,619           Other accrued expenses         6,088         8,539           Total current liabilities         49,399         61,954           Long-term liabilities         159         155           Deferred income taxes         159         155           Long-term liabilities         7,8         122           Total current liabilities         7,8         122           Total tabilities         1,34         1,34           Commitments and contringencies         5         50,984         63,577 <td>Income tax receivable</td> <td></td> <td>_</td>	Income tax receivable		_
Equipment, furniture and fixtures, net         19,90           Goodwill         13,609         13,724           Intangible assets, net         11,527         11,1572           Other assets         757         1,144           Deferred income taxes         7,883         3,833           Total assets         \$225,055         \$233,307           Current liabilities:           Accounts payable         \$0,969         6,028           Accrued alse discounts, rebates and royalties         6,692         6,028           Accrued compensation         5,650         4,619           Other accrued expenses         6,088         8,539           Total current liabilities         15,650         4,619           Long-term Liabilities         15,650         4,619           Long-term Liabilities         15,99         15,38           Deferred income taxes         15,99         15,38           Income taxe payable         15,99         15,38           Other long-term liabilities         7,8         12,22           Commitments and contringencies         15,99         15,31           Stockholders' equity:         2,00         1,00         1,00           Perferred stock, \$0,01 par value, 5,000,	Deferred income taxes	2,938	3,016
Goodwill         13,609         13,724           Intangible assets, net         11,323         11,572           Other assets         757         1,144           Defered income taxes         7,833         7,833           Total assets         \$225,055         \$233,307           LIABILITIES AND STOCKHOLDERS' EQUITY           Current liabilities:           Current liabilities:           Accrued sales discounts, rebates and royalties         6,922         6,028           Accrued compensation         5,050         6,028           Accrued compensation         6,088         8,539           Total current liabilities         49,399         61,954           Deferred income taxes         159         15,96           Income tax payable         1,348         1,348           Other Jong-term liabilities         159         15,27           Deferred income taxes         159         15,28           Other Jong-term liabilities         78         122           Total liabilities         50,984         63,57           Commitments and contingencies           Stockholders' equity:           Preferred income, 5,000,000 shares authorized; 19,270,607 and 19,14	Total current assets	180,600	189,040
Goodwill         13,609         13,724           Intangible assets, net         11,323         11,572           Other assets         757         1,144           Defered income taxes         7,833         7,833           Total assets         \$225,055         \$233,307           LIABILITIES AND STOCKHOLDERS' EQUITY           Current liabilities:           Current liabilities:           Accrued sales discounts, rebates and royalties         6,922         6,028           Accrued compensation         5,050         6,028           Accrued compensation         6,088         8,539           Total current liabilities         49,399         61,954           Deferred income taxes         159         15,96           Income tax payable         1,348         1,348           Other Jong-term liabilities         159         15,27           Deferred income taxes         159         15,28           Other Jong-term liabilities         78         122           Total liabilities         50,984         63,57           Commitments and contingencies           Stockholders' equity:           Preferred income, 5,000,000 shares authorized; 19,270,607 and 19,14	Equipment, furniture and fixtures, net	10,913	9,990
Intangible assets, net         11,323         11,572           Other assets         757         1,144           Deferred income taxes         8,285         \$23,307           Total assets         \$225,055         \$233,307           LIABILITIES AND STOCKHOLDERS' EQUITY           Current liabilities:           Accrued lasel discounts, rebates and royalties         \$ 30,969         \$ 39,514           Accrued ales discounts, rebates and royalties         6,692         6,028           Accrued income taxes         6,692         6,028           Accrued compensation         5,056         4,619           Other accrued expenses         6,088         8,339           Total current liabilities         159         153           Income taxes         159         153           Income, taxe payable         159         153           Other long-term liabilities         159         153           Competential income taxes         5,093         3,241           Other long-term liabilities         5,093         6,357           Commitments and contingencies           Stockholders' equity:           Perfered stock, 9,001 par value, 5,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on Septe	• •		
Other assets         757         1,144           Deferred income taxes         7,853         7,837           Total assets         ELABILITIES AND STOCKHOLDERS' EQUITY           Current liabilities           Accounts payable         \$ 30,969         \$ 39,154           Accrued siscounts, rebates and royalties         6,602         6,028           Accrued compensation         5,650         4,619           Other accrued expenses         6,088         8,539           Total current liabilities         49,399         6,535           Total current liabilities         13         13           Deferred income taxes         1         1           Income tax payable         1,348         1,348           Other long-term liabilities         1         3         1,348           Total current liabilities         5,598         1         1           Total liabilities         5         1         1           Commitments are spayable         1         4         3,48           Other long-term liabilities         5         5,594         1         1           Commitments and contingencies         Commentax labority and contingencies         Commentax labority and con	Intangible assets, net		
Deferred income taxes         7,853         7,835           Total assets         5225,055         5233,307           LIABILITIES AND STOCKHOLDERS' EQUITY           Current liabilities:           Current liabilities:           Accounts payable         6,692         6,028           Accrued cincome taxes         6,692         4,018           Accrued compensation         6,688         8,539           Other accrued expenses         6,088         8,539           Total current liabilities         49,399         61,954           Long-term liabilities         159         153           Income tax payable         1,348         1,348         1,348           Other long-term liabilities         78         122           Total liabilities         50,984         63,572           Commitments and contingencies           Stockholders' equity:           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         —         —           Commitments and contingencies         —         —         —           Stockholders' equity:         —         —         —           Preferred stock, \$0.01 par value,			
Total assets   S235.05   S233.07	Deferred income taxes	7,853	
LIABILITIES AND STOCKHOLDERS' EQUITY           Current liabilities:           Accounts payable         \$ 30,969         \$ 39,514           Accrued sales discounts, rebates and royalties         6,692         6,028           Accrued income taxes         -         3,254           Accrued compensation         5,650         4,619           Other accrued expenses         6,088         8,539           Total current liabilities         49,399         61,954           Long-term liabilities         159         153           Income tax payable         1,348         1,348           Other long-term liabilities         78         122           Total liabilities         50,984         63,577           Commitments and contingencies           Stockholders' equity:           Preferred stock, \$0,01 par value, 5,000,000 shares authorized; none issued or outstanding         -         -           Preferred stock, \$0,01 par value, 5,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on         193         191           September 30, 2010 and December 31, 2009, respectively         133,078         128,913           Accumulated other comprehensive (loss) income         (168)         1,463           Retained earnings <t< td=""><td>Total assets</td><td></td><td></td></t<>	Total assets		
Current liabilities:           Accounts payable         \$ 30,969         \$ 39,514           Accrued sales discounts, rebates and royalties         6,022         6,028           Accrued compensation         5,650         4,619           Other accrued expenses         6,088         8,339           Total current liabilities         49,399         61,954           Deferred income taxes         1,58         1,53           Income tax payable         1,348         1,348           Other long-term liabilities         78         122           Total liabilities         78         122           Commitments and contingencies         5,094         63,577           Commitments and contingencies         5,094         63,577           Stockholders' equity:         Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         —         —           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on         193         191           Paid-in capital         133,078         128,913           Accumulated other comprehensive (loss) income         113,078         128,913           Accumulated earnings         263,407         249,556           Less cost of common stock in treasury, 5,921,	Total assets	<u>#223,033</u>	Ψ233,307
Accounts payable         \$ 30,969         \$ 39,514           Accrued sales discounts, rebates and royalties         6,692         6,028           Accrued income taxes         -         3,254           Accrued compensation         5,650         4,619           Other accrued expenses         6,088         8,539           Total current liabilities         49,399         61,954           Long-term liabilities         159         153           Income tax payable         1,348         1,348           Other long-term liabilities         78         122           Total liabilities         78         122           Total liabilities         50,984         63,577           Commitments and contingencies           Stockholders' equity:           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         —         —           Common stock, \$0.01 par value, 50,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on         193         191           Paid-in capital         133,078         128,913           Accumulated other comprehensive (loss) income         (168)         1,463           Retained earnings         (168)         1,463           Less cost of common stock in treasury, 5,921,			
Accrued sales discounts, rebates and royalties         6,692         6,028           Accrued income taxes         -         3,254           Accrued compensation         6,088         8,539           Other accrued expenses         6,088         8,539           Total current liabilities         49,399         61,954           Long-term liabilities:         159         153           Income tax payable         1,348         1,348           Other long-term liabilities         78         122           Total liabilities         50,984         63,577           Commitments and contingencies         -         -           Stockholders' equity:           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         -         -           Common stock, \$0.01 par value, 5,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on September 30, 2010 and December 31, 2009, respectively         193         191           Paid-in capital         133,078         128,913           Accumulated other comprehensive (loss) income         (168)         1,463           Retained earnings         (168)         1,463           Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively         (89,336)	Current liabilities:		
Accrued income taxes         —         3,254           Accrued compensation         5,650         4,619           Other accrued expenses         6,088         8,539           Total current liabilities         49,399         61,954           Long-term liabilities:         159         153           Income tax payable         1,348         1,348           Other long-term liabilities         78         122           Total liabilities         50,984         63,577           Commitments and contingencies           Stockholders' equity:           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         —         —           Common stock, \$0.01 par value, 5,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on         193         191           September 30, 2010 and December 31, 2009, respectively         193         191           Paid-in capital         133,078         128,913           Accumulated other comprehensive (loss) income         (168)         1,463           Retained earnings         130,304         118,989           263,407         249,556           Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively         (89,336)         <		\$ 30,969	\$ 39,514
Accrued compensation         5,650         4,619           Other accrued expenses         6,088         8,539           Total current liabilities         49,399         61,954           Long-term liabilities:         5         159         153           Income tax payable         1,348         1,348         1,348         1,248         1,22           Total liabilities         78         122         50,984         63,577           Commitments and contingencies           Stockholders' equity:           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         —         —         —           Common stock, \$0.01 par value, 5,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on September 30, 2010 and December 31, 2009, respectively         193         191           Paid-in capital         133,078         128,913         128,913           Accumulated other comprehensive (loss) income         (168)         1,463           Retained earnings         130,304         118,989           Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively         (89,336)         (79,826)           Total stockholders' equity         174,071         169,730         179,826 </td <td>Accrued sales discounts, rebates and royalties</td> <td>6,692</td> <td>6,028</td>	Accrued sales discounts, rebates and royalties	6,692	6,028
Other accrued expenses         6,088         8,539           Total current liabilities         49,399         61,954           Long-term liabilities:         5         153           Deferred income taxes         159         153           Income tax payable         1,348         1,348           Other long-term liabilities         78         122           Total liabilities         50,984         63,577           Commitments and contingencies           Stockholders' equity:           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         —         —           Common stock, \$0.01 par value, 50,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on         193         191           Paid-in capital         133,078         128,913           Accumulated other comprehensive (loss) income         (168)         1,463           Retained earnings         130,304         118,989           Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively         (89,336)         (79,826)           Total stockholders' equity         174,071         169,730         169,730	Accrued income taxes	_	3,254
Total current liabilities         49,399         61,954           Long-term liabilities:         159         153           Deferred income taxes         1,348         1,348           Income tax payable         1,348         1,228           Other long-term liabilities         78         122           Total liabilities         50,984         63,577           Stockholders' equity:           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         —         —           Common stock, \$0.01 par value, 50,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on September 30, 2010 and December 31, 2009, respectively         193         191           Paid-in capital         133,078         128,913           Accumulated other comprehensive (loss) income         (168)         1,463           Retained earnings         130,304         118,989           Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively         (89,336)         (79,826)           Total stockholders' equity         174,071         169,730	Accrued compensation	5,650	4,619
Deferred income taxes   159   153   163	Other accrued expenses	6,088	8,539
Deferred income taxes   159   153   163	Total current liabilities	49,399	61,954
Deferred income taxes   159   153     Income tax payable   1,348   1,348     Other long-term liabilities   78   122     Total liabilities   50,984   63,577     Commitments and contingencies	Long-term liabilities:		
Other long-term liabilities         78         122           Total liabilities         50,984         63,577           Commitments and contingencies           Stockholders' equity:		159	153
Other long-term liabilities         78         122           Total liabilities         50,984         63,577           Commitments and contingencies           Stockholders' equity:	Income tax payable	1,348	1,348
Total liabilities         50,984         63,577           Commitments and contingencies         -         -           Stockholders' equity:         -         -           Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding         -         -           Common stock, \$0.01 par value, 50,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on         193         191           September 30, 2010 and December 31, 2009, respectively         133,078         128,913           Accumulated other comprehensive (loss) income         (168)         1,463           Retained earnings         130,304         118,989           Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31,         (89,336)         (79,826)           Total stockholders' equity         174,071         169,730			
Stockholders' equity:         Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding       —       —         Common stock, \$0.01 par value, 50,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on       193       191         September 30, 2010 and December 31, 2009, respectively       193       128,913         Accumulated other comprehensive (loss) income       (168)       1,463         Retained earnings       130,304       118,989         Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively       (89,336)       (79,826)         Total stockholders' equity       174,071       169,730	-	50,984	63,577
Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding       —       —       —         Common stock, \$0.01 par value, 50,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on       193       191         September 30, 2010 and December 31, 2009, respectively       193       191         Paid-in capital       133,078       128,913         Accumulated other comprehensive (loss) income       (168)       1,463         Retained earnings       130,304       118,989         Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31,       (89,336)       (79,826)         Total stockholders' equity       174,071       169,730	Commitments and contingencies		
Common stock, \$0.01 par value, 50,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on September 30, 2010 and December 31, 2009, respectively       193       191         Paid-in capital       133,078       128,913         Accumulated other comprehensive (loss) income       (168)       1,463         Retained earnings       130,304       118,989         Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively       (89,336)       (79,826)         Total stockholders' equity       174,071       169,730	Stockholders' equity:		
September 30, 2010 and December 31, 2009, respectively       193       191         Paid-in capital       133,078       128,913         Accumulated other comprehensive (loss) income       (168)       1,463         Retained earnings       130,304       118,989         263,407       249,556         Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively       (89,336)       (79,826)         Total stockholders' equity       174,071       169,730	Preferred stock, \$0.01 par value, 5,000,000 shares authorized; none issued or outstanding	_	_
Paid-in capital       133,078       128,913         Accumulated other comprehensive (loss) income       (168)       1,463         Retained earnings       130,304       118,989         263,407       249,556         Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31,       (89,336)       (79,826)         Total stockholders' equity       174,071       169,730	Common stock, \$0.01 par value, 50,000,000 shares authorized; 19,270,607 and 19,140,232 shares issued on		
Accumulated other comprehensive (loss) income       (168)       1,463         Retained earnings       130,304       118,989         263,407       249,556         Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31,       (89,336)       (79,826)         Total stockholders' equity       174,071       169,730	September 30, 2010 and December 31, 2009, respectively		191
Retained earnings       130,304       118,989         263,407       249,556         Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively       (89,336)       (79,826)         Total stockholders' equity       174,071       169,730	Paid-in capital	133,078	128,913
Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively       (89,336) (79,826)         Total stockholders' equity       174,071 169,730	Accumulated other comprehensive (loss) income	(168)	1,463
Less cost of common stock in treasury, 5,921,136 and 5,449,962 shares on September 30, 2010 and December 31, 2009, respectively  Total stockholders' equity  (89,336) (79,826) 174,071 169,730	Retained earnings	130,304	118,989
2009, respectively       (89,336)       (79,826)         Total stockholders' equity       174,071       169,730		263,407	249,556
Total stockholders' equity 174,071 169,730		(89 336)	(79.826)
<u>=====================================</u>	• •		
		=======================================	<del>4 200,007</del>

## UNIVERSAL ELECTRONICS INC. CONSOLIDATED INCOME STATEMENTS

(In thousands, except per share amounts) (Unaudited)

		Three Months Ended September 30,		Nine Months Ended September 30,	
	2010	2009	2010	2009	
Net sales	\$ 79,007	\$ 83,182	\$229,275	\$ 232,611	
Cost of sales	53,289	57,112	154,068	159,609	
Gross profit	25,718	26,070	75,207	73,002	
Research and development expenses	2,687	2,251	7,944	6,411	
Selling, general and administrative expenses	16,465	17,175	50,694	52,724	
Operating income	6,566	6,644	16,569	13,867	
Interest (expense) income, net	(1)	110	99	376	
Other income (expense), net	40	25	62	(161)	
Income before provision for income taxes	6,605	6,779	16,730	14,082	
Provision for income taxes	(1,903)	(2,556)	(5,415)	(5,247)	
Net income	\$ 4,702	\$ 4,223	\$ 11,315	\$ 8,835	
Earnings per share:					
Basic	\$ 0.35	\$ 0.31	\$ 0.83	\$ 0.65	
Diluted	\$ 0.34	\$ 0.30	\$ 0.81	\$ 0.63	
Shares used in computing earnings per share:					
Basic	13,417	13,687	13,572	13,656	
Diluted	13,671	14,008	13,897	13,940	

## UNIVERSAL ELECTRONICS INC. CONSOLIDATED STATEMENTS OF CASH FLOWS

(In thousands) (Unaudited)

	Nine Mon Septen	ths Ended
	2010	2009
Cash provided by operating activities:		
Net income	\$ 11,315	\$ 8,835
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation and amortization	4,660	5,019
Provision for doubtful accounts	661	181
Provision for inventory write-downs	1,914	2,861
Deferred income taxes	17	(297)
Tax benefit from exercise of stock options	102	374
Excess tax benefit from stock-based compensation	(131)	(198)
Shares issued for employee benefit plan	375	530
Stock-based compensation	3,757	3,184
Changes in operating assets and liabilities:		
Accounts receivable	4,508	4,474
Inventories	(6,069)	(1,931)
Prepaid expenses and other assets	1,158	883
Accounts payable and accrued expenses	(8,229)	(3,241)
Accrued income taxes	(3,407)	202
Net cash provided by operating activities	10,631	20,876
Cash used for investing activities:		
Term deposit	(290)	(49,125)
Acquisition of equipment, furniture and fixtures	(4,449)	(4,142)
Acquisition of intangible assets	(1,061)	(988)
Acquisition of assets from Zilog, Inc.	(1,001)	(9,502)
Net cash used for investing activities	(5,800)	(63,757)
-		
Cash used for financing activities:		
Proceeds from stock options exercised	257	2,412
Treasury stock purchased	(9,835)	(5,242)
Excess tax benefit from stock-based compensation	131	198
Net cash used for financing activities	(9,447)	(2,632)
Effect of exchange rate changes on cash	(953)	374
Net decrease in cash and cash equivalents	(5,569)	(45,139)
Cash and cash equivalents at beginning of period	29,016	75,238
Cash and cash equivalents at end of period	\$ 23,447	\$ 30,099